# Using mobile distributed pyrolysis facilities to deliver a forest residue resource for bio-fuel production

by

Duncan Brown BSc (Hons), University of Birmingham, 2009

A Thesis Submitted in Partial Fulfillment of the Requirements for the Degree of

# MASTER OF APPLIED SCIENCE

in the Department of Mechanical Engineering

© Duncan Brown, 2013 University of Victoria

All rights reserved. This thesis may not be reproduced in whole or in part, by photocopy or other means, without the permission of the author.

# **Supervisory Committee**

# Using mobile distributed pyrolysis facilities to deliver a forest residue resource for bio-fuel production

by

Duncan Brown BSc (Hons), University of Birmingham, 2009

# **Supervisory Committee**

Dr. Andrew Rowe (Department of Mechanical Engineering) Co-Supervisor

Dr. Peter Wild (Department of Mechanical Engineering) Co-Supervisor

### Abstract

#### **Supervisory Committee**

Dr. Andrew Rowe (Department of Mechanical Engineering) Co-Supervisor Dr. Peter Wild (Department of Mechanical Engineering) Co-Supervisor

Distributed mobile conversion facilities using either fast pyrolysis or torrefaction processes can be used to convert forest residues to more energy dense substances (bio-oil, bio-slurry or torrefied wood) that can be transported as feedstock for bio-fuel facilities. All feedstock are suited for gasification, which produces syngas that can be used to synthesise petrol or diesel via Fischer-Tropsch reactions, or produce hydrogen via water gas shift reactions. Alternatively, the bio-oil product of fast pyrolysis may be upgraded to produce petrol and diesel, or can undergo steam reformation to produce hydrogen.

Implementing a network of mobile facilities reduces the energy content of forest residues delivered to a bio-fuel facility as mobile facilities use a fraction of the biomass energy content to meet thermal or electrical demands. The total energy delivered by bio-oil, bio-slurry and torrefied wood is 45%, 65% and 87% of the initial forest residue energy content, respectively. However, implementing mobile facilities is economically feasible when large transport distances are required. For an annual harvest of 1.717 million m<sup>3</sup> (equivalent to 2000 ODTPD), transport costs are reduced to less than 40% of the total levelised delivered feedstock cost when mobile facilities are implemented; transport costs account for up to 80% of feedstock costs for conventional woodchip delivery. Torrefaction provides the lowest cost pathway of delivering a forest residue resource when using mobile facilities. Cost savings occur against woodchip delivery for annual forest residue harvests above 2.25 million m<sup>3</sup> or when transport distances greater than 250 km are required.

Important parameters that influence levelised delivered costs of feedstock are transport distances (forest residue spatial density), haul cost factors, thermal and electrical demands of mobile facilities, and initial moisture content of forest residues. Relocating mobile facilities can be optimised for lowest cost delivery as transport distances of raw biomass are reduced.

The overall cost of bio-fuel production is determined by the feedstock delivery pathway and also the bio-fuel production process employed. Results show that the minimum cost of petrol and diesel production is 0.86 \$ litre<sup>-1</sup> when a bio-oil feedstock is upgraded. This corresponds to a 2750 TPD upgrading facility requiring an annual harvest of 4.30 million m<sup>3</sup>. The minimum cost of hydrogen production is 2.92 \$ kg<sup>-1</sup>, via the gasification of a woodchip feedstock and subsequent water gas shift reactions. This corresponds to a 1100 ODTPD facility and requires an annual harvest of 947,000 m<sup>3</sup>.

The levelised cost of bio-fuel strongly depends on the size of annual harvest required for bio-fuel facilities. There are optimal harvest volumes (bio-fuel facility sizes) for each bio-fuel production route, which yield minimum bio-fuel production costs. These occur as the benefits of economies of scale for larger bio-fuel facilities compete against increasing transport costs for larger harvests. Optimal harvest volumes are larger for bio-fuel production routes that use feedstock sourced from mobile facilities, as mobile facilities reduce total transport requirements.

# **Table of Contents**

S	ıper	viso	ry Committeeii
A	bstr	act	iii
T	able	of C	Contents v
L	ist o	f Ta	blesviii
L	st o	f Fig	guresix
N	ome	encla	turexi
1	N	Moti	vation1
	1.1		Climate change, forest residues and bio-fuels 1
	1.2		Objective
	1.3		Outline
2	I	Back	ground 4
	2.1		Forest residues 4
	2.2		Mobile facility processes $\epsilon$
	2	2.2.1	Fast pyrolysis
	2	2.2.2	Torrefaction
	2.3		Bio-fuel production processes11
	2	2.3.1	Gasification
	2	2.3.2	Fischer-Tropsch reactions
	2	2.3.3	Water gas shift reactions14
	2	2.3.4	Upgrading bio-oil14
	2	2.3.5	Steam reformation of bio-oil
	2.4		Summary
3	N	Meth	ods17
	3.1		Defining a forest residue resource
	3.2		Feedstock collection and delivery to bio-fuel facility
	3	3.2.1	Point-of-delivery scenarios
	3	3.2.2	Harvest and transport
	3	3.2.3	Mobile facilities

	3.2.4	Costs of feedstock collection and delivery	. 29
	3.2.5	Summary	. 33
	3.3 E	io-fuel facilities and production	. 33
	3.3.1	Bio-fuel facilities	. 34
	3.3.2	Bio-fuel production	. 36
	3.3.3	Costs of bio-fuel production	. 39
	3.3.4	Summary	. 43
	3.4 A	analysis performed	. 43
	3.5 S	ummary	. 44
4	Result	s and Discussion	. 45
	4.1 E	ase analysis	. 46
	4.1.1	Levelised delivered cost	. 46
	4.1.2	Levelised cost of bio-fuel	. 47
	4.1.3	Validation	. 49
	4.2 S	ensitivity study	. 51
	4.2.1	Levelised delivered cost	. 51
	4.2.2	Levelised cost of bio-fuel	. 52
	4.3 F	eedstock collection and delivery	. 53
	4.3.1	Annual volume of forest residues	. 53
	4.3.2	Forest residue spatial density	. 55
	4.3.3	Initial moisture content	. 55
	4.3.4	Transport	. 56
	4.3.5	Relocation of mobile facilities	. 59
	4.3.6	Point-of-delivery	. 59
	4.4 E	Sio-fuel production	. 62
	4.4.1	Petrol and diesel	. 62
	4.4.2	Hydrogen	. 67
	4.5 C	Other considerations	. 70
	4.5.1	Forest residue resource	. 70
	4.5.2	Market demand and marginal costs	. 70
	4.5.3	Mobile facility configurations	. 71

vi

v	ii	
4.5.4 Sustainability of bio-fuel production from forest residues	1	
4.5.5 Initiating use of mobile facilities7	2	
4.6 Summary7	3	
5 Conclusions	6	
5.1 Recommendations	7	
6 References	9	
Appendix A: Additional calculations		
Appendix B: Input data	2	
Appendix C: Sensitivity study results	7	

# List of Tables

Table 2-1	Conversion factors between units used to measure forest residue resource	
	quantities (moisture content 50% wt).	6
Table 4-1	Cost categories	. 45
Table 4-2	Breakdown of levelised delivered cost (LDC) for base harvest	. 47
Table 4-3	Validation of the model against results available in the literature.	. 50
Table 4-4	Levelised delivered costs for a range of truck types and haul costs	. 58
Table 4-5	Levelised costs of bio-fuels via upgrading bio-oil using hydrogen purchase	ed
	from an external source	. 65
Table B-1	Forest residue resource input data (technical)	. 92
Table B-2	Forest residue resource input data (financial)	. 92
Table B-3	Transport input data (technical)	. 93
Table B-4	Transport input data (financial)	. 93
Table B-5	Mobile facility input data (technical)	. 94
Table B-6	Mobile facility input data (financial)	. 94
Table B-7	Bio-fuel facility input data (technical)	. 95
Table B-8	Bio-fuel facility input data (financial)	. 96
Table C-1	Sensitivity study results for levelised delivered cost	. 97
Table C-2	Sensitivity study results for levelised cost of bio-fuel (Gasification and	
	Fischer-Tropsch bio-fuel production route)	. 98
Table C-3	Sensitivity study results for levelised cost of bio-fuel (Gasification and wa	ter
	gas shift bio-fuel production route)	. 98
Table C-4	Sensitivity study results for levelised cost of bio-fuel (Upgrading bio-oil	
	bio-fuel production route)	. 98
Table C-5	Sensitivity study results for levelised cost of bio-fuel (Steam reformation of	of
	bio-oil bio-fuel production route)	. 98

# List of Figures

Figure 2-1	Typical auger reactor setup7
Figure 2-2	Overview of auger fast pyrolysis and process yields7
Figure 2-3	Overview of torrefaction and process yields10
Figure 2-4	Schematic of BCL (Silvagas) gasification reactor
Figure 2-5	Bio-oil upgrading processes and hydrogen source options15
Figure 3-1	Four pathways for delivering a forest residue resource
Figure 3-2	Point-of-delivery scenarios
Figure 3-3	a) Harvest and transport model for delivering a conventional woodchip feedstock.
	b) Overview of harvest grid for harvest and transport model when
	implementing mobile facilities.
	c) A distributed collection region with mobile facility at the centre.
	d) Average transport distances (scenario A)
Figure 3-4	Bio-fuel production options
Figure 4-1	a) Net present value of costs for all bio-fuel production routes.
	b) Levelised cost of bio-fuel for all bio-fuel production routes
Figure 4-2	a) Levelised delivered cost over a range of annual harvest volumes.
	b) Cost components for a bio-fuel facility of size 500 ODTPD
	(woodchip equivalent).
	c) Cost components for a bio-fuel facility of size 5000 ODTPD
	(woodchip equivalent)
Figure 4-3	Levelised delivered costs for a range of forest residue spatial densities 56
Figure 4-4	a) Levelised delivered costs for a range of initial moisture content of forest residues.
	b) Daily propane requirements at mobile facilities for a range of drying
	efficiencies and initial moisture content of forest residues
Figure 4-5	Variation in levelised delivered cost of torrefied wood when relocating
	mobile torrefaction facilities

Figure 4-6	Levelised delivered cost when additional transport distances to a bio-fuel
	facility are required (scenario B; annual harvest of 1.717 million m <sup>3</sup> ) 61
Figure 4-7	Visualisation of lowest cost delivery option for a range of harvest volumes
	and additional transport distances
Figure 4-8	Levelised cost of bio-fuel (petrol and diesel) for a range of annual
	harvest volumes
Figure 4-9	Levelised cost of bio-fuel (petrol and diesel) for optimally sized
	bio-fuel facilities when additional transport distances to a bio-fuel facility
	are required (scenario B)
Figure 4-10	Levelised cost of bio-fuel (petrol and diesel) for ranges of hydrogen
	requirements for upgrading and hydrogen production from bio-oil steam
	reformation provided in the literature
Figure 4-11	Levelised cost of bio-fuel (hydrogen) for a range of annual harvest
	volumes
Figure 4-12	Levelised cost of bio-fuel (hydrogen) for optimally sized bio-fuel facilities
	when additional transport distances to a bio-fuel facility are required
	(scenario B)
Figure A-1	Diesel consumption rates for a 200 kW generator

# Nomenclature

## Acronyms

BCL	Battelle Columbus Laboratory
CRF	Capital recovery factor
FT	Fischer-Tropsch
GT	Green tonne
NPV	Net present value
ODT	Oven-dry tonne
ODTPD	Oven-dry tonne per day
ODW	Oven-dry wood
ROI	Renewable Oil International
SMR	Steam Methane Reformation
WGS	Water gas shift

# Symbols

Note: all lower case symbols represent values or data input into the model and all upper case symbols represent values or data calculated by the model

Ci	Unit cost of $i$ (\$·[unit] <sup>-1</sup> )
C <sub>mob</sub>	Cost of mobile facility (\$)
C <sub>ref,k</sub>	Reference cost of bio-fuel facility $k$ (\$)
C <sub>ref,tank</sub>	Reference cost of stainless steel tank (\$)
C <sub>tran,fix,j</sub>	Unit fixed cost of transportation for commodity $j$ (\$-tonne <sup>-1</sup> )
C <sub>tran,var,j</sub>	Unit variable cost of transportation for commodity $j$ ( $\cdot km^{-1}$ )
d	Additional transport distance (km)
e <sub>dry</sub>	Energy required to dry woodchips (MJ·[kg-water-removed] <sup>-1</sup> )
$f_k$	Ratio of bio-fuel yield per unit feed at bio-fuel facility $k$
f <sub>r,k</sub>	Petrol-diesel production ratio at bio-fuel facility $k$

f <sub>smr</sub>	Ratio of hydrogen production from methane
h	Annual quantity of available forest residues $(m^3 \cdot yr^{-1})$
$l_{CH_4}$	Lower heating value of methane (MJ·kg <sup>-1</sup> )
$l_{C_3H_8}$	Lower heating value of propane (MJ·kg <sup>-1</sup> )
l <sub>odw</sub>	Lower heating value of oven-dry wood (MJ·kg <sup>-1</sup> )
$m_{meth}$	Mass ratio of methanol added to bio-oil
$m_{WS}$	Mass fraction of bio-oil that is water soluble
n <sub>r</sub>	Number of relocations for each mobile facility during operation lifetime
n <sub>staff</sub>	Number of staff required at each mobile facility
$O_{bf,k}$	Operation and maintenance percentage of bio-fuel facility $k$
0 <sub>mob</sub>	Operation and maintenance percentage of mobile facilities
0 <sub>tank</sub>	Operation and maintenance percentage of stainless steel tank
s <sub>mob</sub>	Size of mobile facilities (ODTPD)
S <sub>ref,k</sub>	Reference size of bio-facility $k$ (TPD)
S <sub>ref,tank</sub>	Reference size of stainless steel tank
x <sub>i</sub>	Mobile facility mass yield ratio for product <i>i</i>
у	Harvest operation lifetime (yr)
$y_i$	Mobile facility energy yield ratio for product <i>i</i>
Ζ	Moisture content of forest residues (% wt)
Z <sub>ODW</sub>	Moisture content of oven-dry wood (% wt)
$C_{i,bf}$	Cost <i>i</i> relating to bio-fuel facilities
C <sub>i,mob</sub>	Cost <i>i</i> relating to mobile facilities
$C_p$	Cost of process $p$ ( $\cdot yr^{-1}$ )
C <sub>tran,fix,j</sub>	Annual fixed cost of transportation for commodity $j$ ( $\cdot yr^{-1}$ )
C <sub>tran,var,j</sub>	Annual variable cost of transportation for commodity $j$ ( $\$ \cdot yr^{-1}$ )
CRF	Capital recovery factor
$\overline{D}_j$	Average annual transport distance of commodity $j$ (km)
$D_j$	Total annual transport distance of commodity $j$ (km)

$\overline{D}_{F,B}$	Average annual transport distance from logging fields to bio-fuel facility
$\overline{D}_{F,M}$	Average annual transport distance from logging fields to mobile facilities (km)
$\overline{D}_{M,B}$	Average annual transport distance from mobile facilities to bio-fuel facility (km)
E <sub>dry</sub>	Annual energy required to dry woodchips $(GJ \cdot yr^{-1})$
E <sub>i,gross</sub>	Total annual energy content of mobile facility product $i$ (GJ·yr <sup>-1</sup> )
E <sub>j,net</sub>	Total annual energy content of commodity <i>j</i> for bio-fuel production $(GJ \cdot yr^{-1})$
$E_{FR}$	Annual energy content of forest residue resource $(GJ \cdot yr^{-1})$
$F_{kg,j,k,m}$	Quantity of fuel $m$ produced from bio-fuel facility $k$ using feedstock commodity $j$ (kg)
H <sub>mob</sub>	Quantity of available forest residues within distributed collection region (m <sup>3</sup> )
$LCB_{j,k,m}$	Levelised cost of bio-fuel <i>m</i> produced at bio-fuel facility <i>k</i> using commodity feedstock <i>i</i> ( $\$$ ·GJ <sup>-1</sup> )
LDC <sub>i</sub>	Levelised delivered cost of commodity $j$ (\$ $\cdot$ GJ <sup>-1</sup> )
M <sub>diesel</sub>	Annual consumption of diesel at mobile torrefaction facility (tonne $yr^{-1}$ )
M <sub>i,gross</sub>	Annual quantity of mobile facility product $i$ (tonne·yr <sup>-1</sup> )
M <sub>j,feed</sub>	Annual quantity of commodity $j$ used as bio-fuel feedstock (tonne yr <sup>-1</sup> )
M <sub>j,net</sub>	Annual quantity of commodity <i>j</i> transported to bio-fuel facility (tonne· yr <sup>-1</sup> )
M <sub>truck,j</sub>	Load of truck transporting commodity $j$ (tonne)
M <sub>BC,excess</sub>	Annual quantity of excess bio-char not added to bio-slurry (tonne $yr^{-1}$ )
M <sub>BC,net</sub>	Annual quantity of bio-char added to bio-slurry (tonne·yr <sup>-1</sup> )
$M_{CH_4}$	Annual quantity of methane required for SMR (tonne $yr^{-1}$ )
$M_{C_3H_8}$	Annual quantity of propane required to assist drying woodchips (tonne $yr^{-1}$ )
M <sub>GT</sub>	Annual quantity of available forest residues in GT (tonne·yr <sup>-1</sup> )
$M_{H_2O}$	Annual quantity of water removed from woodchips (tonne $yr^{-1}$ )
$M_{H_2,UG}$	Annual quantity of hydrogen required for upgrading (tonne $yr^{-1}$ )
M <sub>ODT</sub>	Annual quantity of available forest residues in ODT (tonne yr <sup>-1</sup> )

N <sub>grid</sub>	Number of distributed collection regions when using mobile facilities
N <sub>mob</sub>	Number of mobile units required to process an annual harvest $h$
N <sub>tank</sub>	Number of liquid storage tanks required at bio-fuel facility
$NPV_j$	Net present value of all costs expended to deliver commodity $j$ (\$)
$NPV_{j,k,m}$	Net present value of all costs expended to produce bio-fuel $m$ at bio-fuel
	facility <i>k</i> using commodity feedstock <i>j</i> (\$)
R	Radius of harvest region (km)
$S_{bf,j,k}$	Size of bio-fuel facility $k$ using commodity feedstock $j$ (TPD)
$V_j$	Annual volume of commodity $j$ delivered to bio-fuel facility (m <sup>3</sup> )
$V_{j,bf}$	Maximum storage volume of commodity $j$ at bio-fuel facility (m <sup>3</sup> )
V <sub>j,mob</sub>	Maximum storage volume of commodity $j$ at each mobile facility (m <sup>3</sup> )
$X_{BO}$	Fraction of gross bio-oil production used for electricity generation

## Greek

Note: all lower case symbols represent values or data input into the model and all upper case symbols represent values or data calculated by the model

β	Electricity demand of mobile facilities (GJ·ODT <sup>-1</sup> )
$\gamma_{H_2}$	Ratio of hydrogen required per unit bio-oil feed for upgrading
ζ <sub>max</sub>	Maximum bio-char loading in bio-slurry (%wt)
$\eta_{dry}$	Efficiency of drying woodchips
$\eta_{gen}$	Efficiency of electricity generator
κ	Cost scaling factor
$ ho_j$	Density of commodity $j$ (kg·m <sup>-3</sup> )
$ ho_{C_3H_8}$	Density of propane $(kg \cdot m^{-3})$
$ ho_D$	Density of diesel (kg $\cdot$ m <sup>-3</sup> )
$ ho_{H_2O}$	Density of water $(kg \cdot m^{-3})$
$ ho_{ODW}$	Density of oven-dry wood $(kg \cdot m^{-3})$
$\sigma_{bf,k}$	Capacity factor of bio-fuel facility k

$\sigma_{mob}$	Maximum capacity factor of mobile facilities
τ	Tortuosity of road network
arphi	Density of available forest residues (m <sup>3</sup> ·km <sup>-2</sup> )
$\gamma_{BO,SR}$	Fraction of bio-oil to be steam reformed during upgrading
Ζ	Bio-char loading in bio-slurry (%wt)

# Subscripts

bf	Relating to bio-fuel facility
i	Mobile facility product <i>i</i> (syngas, bio-oil, bio-char, or torrefied wood)
j	Commodity <i>j</i> for bio-fuel production (woodchip, bio-oil, bio-slurry, or
	torrefied wood)
k	Bio-fuel facility $k$ (FT, WGS, UG or SR)
kg	Kilogram
lab	Labour
т	Bio-fuel $m$ (petrol, diesel or hydrogen)
$m^3$	Cubic metres
meth	Methanol
mob	Relating to mobile facility
p	Process <i>p</i> (e.g. piling, chipping, loading)
ref	Reference value
relocate	Relating to the relocation of mobile facilities
tank	Relating to bio-oil/bio-slurry storage tank
wage	Wage of mobile facility employees
BC	Bio-char
ВО	Bio-oil
BS	Bio-slurry
$CH_4$	Methane
$C_3H_8$	Propane
D	Diesel
FR	Forest residue

FT	Gasification and Fischer-Tropsch facility
Н	Hydrogen
0&M	Operation and maintenance
Р	Petrol
S	Syngas
SMR	Steam methane reformation facility
SR	Bio-oil steam reformation facility
TW	Torrefied wood
UG	Upgrading bio-oil facility
WC	Woodchip
WGS	Water gas shift facility
WS	Water soluble

## 1 Motivation

#### 1.1 Climate change, forest residues and bio-fuels

Climate change concerns and government policies aimed at reducing greenhouse gas emissions from fossil fuels continue to contribute to an increasing demand for fuels from biomass sources ('bio-fuels'). The extent to which bio-fuels mitigate climate change by reducing greenhouse gas emissions compared against fossil fuels is a subject of debate, particularly when land use changes are considered in lifecycle analyses [1]–[5]. However, bio-fuels produced from waste wood, such as forest or mill residuals, can reduce net carbon emissions whilst avoiding controversy related to land use change as they are a by-product of the forest industry [6].

Forest residuals, in particular, have considerable potential for increased utilisation for bio-fuel production - at present, most are burned on-site at the end of commercial forestry operations. However, forest residues suffer from low spatial and energy densities, which hinder their use as a biomass resource. Typically, forest residues are spread-out over wide areas of land, thus large distances are travelled for collection and delivery to bio-fuel production facilities. If forest residues are transported in their raw form or as woodchips, truck capacity is limited by volume rather than weight and, as a result, more delivery trips are required than if the truck were transporting a more energy dense substance at full weight capacity [7]. The combination of low spatial and energy densities of biomass results in high transport costs which, in turn, elevate the final bio-fuel production cost.

One proposed method of reducing the cost of delivering a forest residue resource is to implement a network of distributed biomass conversion facilities near the location of forest residues [8]. These conversion facilities convert raw biomass to a more energy dense substance, which is then transported longer distances to a centralised bio-fuel production facility. Mobile facilities are of particular interest as forest residues are not consistently available at the same location for long periods of time. Mobile distributed conversion facilities ('mobile facilities') can be moved from a depleted region and relocated to a region with abundant forest residues. Relocating mobile facilities reduces transport distances of raw biomass material.

Two processes that are suited for mobile facilities are fast pyrolysis and torrefaction ([9], [10]). These are both forms of pyrolysis, which is the thermal decomposition of materials in the absence of oxygen. Fast pyrolysis involves high heating rates and short reaction times producing primarily a liquid bio-oil product [11]. Torrefaction occurs at lower temperatures than fast pyrolysis and the principal product is a solid char-like substance known as torrefied wood [12]. The energy and mass densities of the liquid and solid products are typically higher than that of forest residues or woodchips.

Upon delivery to a bio-fuel facility, the products of mobile facilities can be used directly as feedstock for bio-fuel production. All are suited for gasification ([11], [13]), which produces syngas that can be used to synthesise petrol or diesel via Fischer-Tropsch reactions, or produce hydrogen via water gas shift reactions. Alternatively, the bio-oil product of fast pyrolysis may be upgraded to produce petrol and diesel, or can undergo steam reformation to produce hydrogen.

Therefore, implementing mobile facilities introduces new pathways for bio-fuel production using a forest residue resource. No literature has been identified that investigates harvesting a forest residue resource using multiple mobile facilities, or the subsequent cost of bio-fuel production when implementing a network of mobile facilities.

#### 1.2 Objective

The objective of this study is to investigate the technical and economic implications of producing bio-fuels when using a network of mobile facilities to deliver a forest residue resource to a bio-fuel facility. A model is created that considers the production of petrol and diesel (for use in present day fuel infrastructure), or hydrogen (which may be a prominent fuel in the near future). The two main outputs of the model are: (i) a levelised delivered cost of feedstock to a bio-fuel facility, and (ii) a final levelised cost of bio-fuel production. The model results are compared against current bio-fuel production methods to determine the feasibility of implementing bio-fuel production pathways using mobile facilities.

### 1.3 Outline

The study is divided into five Chapters. Chapter 2 provides background on forest residue resources, mobile facility processes and bio-fuel production processes. Chapter 3 describes the model constructed to analyse bio-fuel production pathways when implementing mobile facilities to harvest a forest residue resource. Chapter 4 presents model results and a discussion of the technical and economic feasibility of implementing mobile facilities. Conclusions and recommendations of further work are made in Chapter 5.

# 2 Background

This Chapter contains background information on forest residue resources, mobile facilities, and bio-fuel production facilities.

Characteristics of forest residue resources vary depending on location, forest type (tree species) and logging industry practices [14], [15]. This study considers a forest residue resource originating from a temperate forest region. No particular geographic location or logging practice is assumed, which would limit the scope of the study, and therefore a general description of forest residue resources is provided in Section 2.1.

Fast pyrolysis and torrefaction are processes suited to mobile application and are discussed in Section 2.2. The discussion includes technical information about each process, as well as information on associated products and commodities derived from these processes, which can be used as feedstock for bio-fuel production. The current status of mobile facilities is also addressed, with focus on companies based in North America.

The bio-fuel production processes considered in this study are gasification followed by either Fischer-Tropsch synthesis or water gas shift reactions, and upgrading or steam reformation of bio-oil. Fischer-Tropsch and water gas shift processes are selected because these are mature processes that are currently available and applicable to a wide range of feedstock. Upgrading and steam reformation of bio-oil are areas of current research ([16]–[19]) and, although no large scale facilities are known to be operating, these processes are included as they relate directly to the use of mobile fast pyrolysis facilities. Background information on each bio-fuel production process is addressed in Section 2.3.

#### 2.1 Forest residues

The forest residue resource considered in this study is a by-product of conventional logging operations, and has been identified as a biomass resource with potential for increased utilisation [14], [15], [20]. Forest residues are tree tops and limbs, and other off-cuts, that are removed during roundwood harvest and discarded within the

logging fields. They are either left scattered across logged areas or piled at the roadside, where they may be burnt to prevent forest fires at a later time.

Forest residues typically have a moisture content between 23 and 50 % ([21], [22]), resulting in lower heating values ranging between 14 and 8 MJ kg<sup>-1</sup>, respectively [23]. Forest residues also have an ash content of up to 3% wt due to higher portions of bark in the wood, and may become contaminated with soil and other impurities during collection processes [24].

The volume of forest residues that are produced during logging varies, but is approximately 20% of the logged roundwood volume [14], [15]. Forest residue recovery is influenced by biophysical, technical, and economic factors. The biophysical resources that contribute to forest growth (e.g. topography, climate geology, and hydrology) impact the ecologically sustainable level of harvesting forest residues, and thus provide a baseline for forest residue recovery (some jurisdictions define a proportion of forest residues that should be retained) [14]. Additionally, it may only be technically possible to recover 41 - 75% of forest residues [25], and the cost-effectiveness of recovery will depend on a range of economic factors, such as market demand for resource utilisation and marginal costs of recovery [26]–[28], that are outside the scope of this study.

Units used to measure quantities of forest residues varies in the literature. A volume may be stated in cubic metres or, alternatively, a mass may be provided for wet (green) or dry tonnes of residues. The stated quantity of resource will depend on the moisture content of forest residues when using units of cubic metres (m<sup>3</sup>) or green tonnes (GT). Therefore, a unit of oven-dry tonnes (ODT), corresponding to a moisture content of approximately 10% wt, is often used as standard to allow comparison between different feedstock types. However, moisture content and volume are important factors when considering the delivery of a biomass resource and therefore this study uses cubic metres to quantify forest residue resources. Table 2-1 provides conversion factors between units for a forest residue moisture content of 50% wt. See Section 3.1 for details.

m <sup>3</sup> per green tonne	1.62
green tonne per oven-dry tonne	1.61
m <sup>3</sup> per oven-dry tonne	2.61

 Table 2-1 Conversion factors between units used to measure forest residue resource quantities (moisture content 50% wt).

#### 2.2 Mobile facility processes

#### 2.2.1 Fast pyrolysis

Pyrolysis is the thermal decomposition of materials in the absence of oxygen [29]. The products of pyrolysis reactions comprise solid char, liquid bio-oil and syngas, which is a mixture of hydrogen, carbon monoxide, carbon dioxide, and methane. The relative quantity and composition of each product depend on operating conditions, such as reaction temperature, residence time, use of catalysts, as well as the feedstock type.

To obtain a high liquid product yield, fast pyrolysis is used. Fast pyrolysis requires temperatures of 500 - 650°C and feed particles less than 3 mm in diameter [29], [30]. The feed particles are resident in the reactor for 2 to 3 seconds, and the vapours produced are condensed into bio-oil. High bio-oil yields of up to 80 wt% can be achieved [31].

There are many types of reactor systems that perform fast pyrolysis including fluid bed, circulating fluid bed, auger, rotating cone and ablative, of which the first three have a strong technical grounding and are most attractive for commercial development [32]. Auger pyrolysis has been selected as the reactor for this study because it has been suggested as an option for mobile facilities [10].

Auger fast pyrolysis requires a feed composed of granules less than 3 mm in diameter with a moisture content of approximately 10%. The feed particles are passed into the reactor where they are indirectly heated to approximately 500°C within 1 second, and the vapours produced are rapidly condensed into bio-oil. The feed system and reactor contain a screw conveyor to enable the process to run continuously (Figure 2-1). It is possible for the non-condensable syngas to be recycled within the reactor to provide process energy, and the solid char exits the reactor vessel for collection [10].



Figure 2-1 Typical auger reactor setup [33].



**Figure 2-2** Overview of auger fast pyrolysis and process yields [10]. The energy content ratios for each product have been calculated to ensure typical lower heating values are maintained (syngas 8  $MJ \cdot kg^{-1}$ ; bio-oil 16.2  $MJ \cdot kg^{-1}$ ; bio-char 19  $MJ \cdot kg^{-1}$ ).

The typical yield of bio-oil tends to be lower for auger reactors compared to fluid bed reactors as the vapours spend more time in the reactor vessel and secondary thermal break-down reactions occur (in contact with the solid char) reducing the quantity of vapours that condense into bio-oil [11]. The yields of auger pyrolysis used in this study are 57% wt bio-oil, 26% wt char and 15% wt syngas (Figure 2-2). Ash is assumed to comprise 2% wt of the products, and exits the reactor along with the solid char. It is possible for the electrical and thermal energy requirements of a mobile fast pyrolysis facility to be supplied using fractions of the pyrolysis products. Thermal demands can be met using syngas, and electrical demands can be met by using bio-oil to power a generator. In cases of high initial moisture content, propane may be required for sufficient drying of biomass [10].

No mobile fast pyrolysis facilities have yet been commercialised, although there are a number of companies working on designs, such as Agri-Therm (Ontario, Canada) and Renewable Oil International (ROI) (Alabama, United States). ROI has manufactured a 5 oven-dry tonne per day (ODTPD) mobile facility and a 15 ODTPD fixed facility, and has plans to construct larger facilities [34], [35]. A 50 ODTPD mobile auger fast pyrolysis design by ROI was the focus of a recent study and is the subject of this analysis [10]. This 50 ODTPD facility is permanently mounted on two 16 metre lowboy trailers, for ease of mobilisation.

#### 2.2.1.1 Bio-oil

Fast pyrolysis liquid, known as bio-oil, has a lower heating value of 14 to 19 MJ kg<sup>-1</sup> and a density of approximately 1200 kg m<sup>-3</sup> [11], [36]. The properties of the bio-oil are strongly influenced by the feed used for the fast pyrolysis process. Bio-oil is often dark-brown in colour, and is a free-flowing heterogeneous mixture composed primarily of oxygenated hydrocarbons and water. The reactive oxygenated compounds such as acids, ethers, esters, aldehydes, ketones and alcohols cause undesirable properties including high viscosity, low pH, immiscibility with fossil fuels, thermal instability and a tendency to polymerize under exposure to air [11], [37]. Therefore, removing the oxygen content of bio-oil, in a process known as upgrading, is often required although it is possible to use raw bio-oil as fuel in a diesel or flex fuel generator [10].

Bio-oil is often stored in stainless steel tanks due to the corrosive nature of the liquid [8]. Secondary reactions within the bio-oil can occur over time in a process known as aging, which results in increased viscosity and in some cases, phase separation. The aging process is accelerated by the presence of fine char within the liquid, but can be reduced by the addition of alcohols such as ethanol or methanol [11]. Downstream

options for utilisation of bio-oil include: electricity generation, bio-fuel production via gasification or upgrading, steam reformation to hydrogen, and production of chemicals.

#### 2.2.1.2 Bio-slurry

Char produced from the fast pyrolysis reaction can be added to bio-oil up to 20% wt to form a bio-slurry mixture that is flowable [38], [39]. Even low quality bio-oils, which are prone to phase separation and contain char and ash contaminates, are amenable to bio-slurry preparation. Adding bio-char to the bio-oil can allow up to 90% of the energy content of the pyrolysis products to be contained in the bio-slurry [40], making bio-slurry an attractive energy carrier for biomass. The density and energy content of bio-slurry varies depending on the char loading in the mixture, but is approximately 1300 kg m<sup>-3</sup> and 30 MJ kg<sup>-1</sup> respectively for a slurry containing 20% wt char. Bio-slurry can be easily pumped into high pressurised gasifiers or other processing reactors to generate electricity or produce bio-fuels [38].

#### 2.2.2 Torrefaction

Torrefaction is a mild pyrolysis process that requires temperatures of 200 - 300°C to decompose the hemicellulose fraction of wood, creating a charcoal-like substance known as torrefied wood [12]. The process begins with initial drying of the woodchips to a moisture content of approximately 10%. Torrefaction occurs when the temperature rises above 200°C. The heating rate of the process remains relatively low (<50°C per min) [13]. Traditionally residence times are up to one hour, although at temperatures of 250 - 280°C it has been shown that residence times as low as 8 minutes can provide torrefied wood with desirable fuel characteristics and grindability [13].

Figure 2-3 shows typical mass and energy balances of the torrefaction process. In general, 70% of the mass remains in the solid product, which contains up to 90% of the input biomass energy. The syngas produced by torrefaction can be used to meet thermal demands, depending on the reactor configuration. Propane may be required to assist with drying biomass, and an external electricity supply may be required to power the process.

Commercial development of torrefaction has not yet been achieved, although auger torrefaction technology has proved a more popular reactor choice for development [41]. In North America, auger torrefaction technology has been designed and built at North Carolina State University. Using this technology, Agri-Tech Producers, LLP (South Carolina, United States) has recently been granted a patent and has plans to design mobile torrefaction facilities (Hopkins and Burnette, "Autothermal and mobile torrefaction devices," US Patent 8 304 590, April 3, 2009). Integro Earth Fuels, also based in North Carolina, has constructed a 100 ODTPD fixed site pilot facility in Gramling, South Carolina [42]. Renewable Fuel Technologies are a California-based company that have begun designs for a 25 ODTPD mobile torrefaction facility [9].

No mobile torrefaction facilities are currently available, thus the technical parameters for torrefaction used in this study have been taken from the literature [13], [43]. A 50 ODTPD facility is assumed with the same initial cost as a mobile fast pyrolysis facility of an equivalent size, given that auger technology is used in both.

#### 2.2.2.1 Torrefied wood

Torrefied wood has a higher energy density than raw biomass feedstock due to the removal of water [43]. The mass density of torrefied wood is lower than raw biomass feedstock, typically 180 - 300 kg m<sup>-3</sup>, as the porosity is increased compared to that of the initial biomass. Torrefied wood is also more brittle than raw biomass, resulting in decreased mechanical strength, making it easier to grind or pulverise [43]. The chemical properties of torrefied wood are similar regardless of the source wood [13], and typical lower heating values range between 18 - 23 MJ kg<sup>-1</sup>. This uniformity of product is an



**Figure 2-3** Overview of torrefaction and process yields [13]. The energy content ratios for each product have been calculated to ensure a typical lower heating values are maintained (syngas  $8 \text{ MJ} \cdot \text{kg}^{-1}$ ; torrefied wood 22.9  $\text{MJ} \cdot \text{kg}^{-1}$ ).

advantage for downstream processes using torrefied wood as an input, such as combustion or gasification. Furthermore, torrefied wood is hydrophobic and therefore more suitable for long term storage than fresh woodchips as fungal degradation is less likely [44].

#### 2.3 Bio-fuel production processes

Woodchips, bio-oil, bio-slurry and torrefied wood are all suited for gasification [11], [13]. Gasification produces syngas that can be used to synthesise petrol or diesel via Fischer-Tropsch reactions, or hydrogen via the water gas shift. Alternatively, bio-oil from fast pyrolysis may be upgraded to produce petrol or diesel, or can undergo steam reformation to produce hydrogen.

#### 2.3.1 Gasification

Gasification is the conversion of carbon-based feedstock into large quantities of gaseous product and small amounts of char and ash [45]. It requires high temperatures between 500 - 1400°C, and may be performed at pressures of 0.1 - 3.3 MPa [46]. The produced gas is cleaned to remove particulates, tars, alkali compounds, sulphur compounds, nitrogen compounds and other contaminants to yield a clean syngas consisting of carbon monoxide, hydrogen, carbon dioxide, water, and methane [47].

Syngas is a 'platform chemical' that can be used for many different purposes, including Fischer Tropsch synthesis of liquid fuels (Section 2.3.2) and hydrogen production via the water gas shift (Section 2.3.3) [38]. Principal reactions (using methane as an example) that occur during gasification to produce syngas are outlined in Equations 2.1 to 2.7:

Reforming:

$$CH_4 + H_2 0 \leftrightarrow CO + 3H_2 \tag{2.1}$$

$$CH_4 + CO_2 \leftrightarrow 2CO + 2H_2 \tag{2.2}$$

Combustion:

$$2CH_4 + O_2 \to 2CO + 4H_2$$
 (2.3)

 $CH_4 + 2O_2 \to CO_2 + 2H_2O$  (2.4)

Water gas shift:

$$CO + H_2O \leftrightarrow CO_2 + H_2 \tag{2.5}$$

Carbon:

$$CH_4 \to C + 2H_2 \tag{2.6}$$

$$2CO \to C + CO_2 \tag{2.7}$$

Reforming reactions (Equation 2.1 and 2.2) are endothermic and require an input of energy provided by concurrent exothermic reactions of partial oxidation (Equation 2.3) or complete combustion (Equation 2.4). The water gas shift reaction (Equation 2.5) is mildly exothermic, and can assist in producing a hydrogen rich syngas under particular reactor conditions. Carbon is also produced during gasification (Equation 2.6 and 2.7).

The three main types of gasification processes are fixed bed, fluidized bed and entrained flow gasifiers [48]. Gasification is an endothermic process and heat may be provided directly (by combustion of the gasification mixture) or indirectly (from an external source - usually a hot solid such as sand or olivine circulating between the gasifier and the char combustor). Each type of gasification process may use steam, air and/or oxygen as a gasification agent to promote conversion. Direct gasification usually uses high pressure air or oxygen, and indirect gasification usually uses steam [48].

This study assumes an entrained flow indirectly heated gasification process designed by Battelle Columbus Laboratory (BCL), commercially known as Silvagas (Figure 2-4). This gasification process was selected as it is applicable to, and has been studied for, biomass based feeds and there is technical and cost information of the process available in the literature [49], [50]. The Silvagas process uses hot sand fluidised by steam to indirectly heat the carbonaceous feedstock to provide the thermal energy required for gasification [46]. The hot sand and char are separated from the gaseous stream into a char combustor where the sand is re-heated and re-circulated back into the gasification reactor. This gasification process produces high quality syngas with minimal nitrogen content [46].

Gasification of bio-oil, bio-slurry or torrefied wood requires a similar process to that of woodchip gasification [51]–[53]. Feeding liquid into a gasification reactor is typically easier than using a feed-hopper technique often used for woodchip feeds [11],



Figure 2-4 Schematic of BCL (Silvagas) gasification reactor [55].

and is an advantage of bio-oil and bio-slurry feeds. In addition, syngas quality is likely to be superior from gasification of bio-oil as many metals and minerals found in raw biomass, which can cause catalyst performance issues, are deposited in the char produced by fast pyrolysis reactions [54]. Gasification of torrefied wood is also found to produce improved syngas quality due to the lower moisture content of torrefied wood [53].

#### 2.3.2 Fischer-Tropsch reactions

Synthetic hydrocarbon fuels can be produced from syngas via Fischer-Tropsch (FT) synthesis reactions. Pressures of 2 - 4 MPa are required [49], and different products are produced when reactions occur at different temperatures. Low temperature synthesis (180 - 250 °C) produces primarily waxes and diesel, and high temperature synthesis

(300 - 350 °C) produces alkenes and petrol [49]. The fuel mixture that is produced requires distillation to separate out petrol and diesel products [49].

The stoichometry of all FT reactions can be represented using two basic reaction equations to describe the production of alkanes (Equation 2.8) and alkenes (Equation 2.9) [56]:

$$nCO + (2n+1)H_2 \leftrightarrow C_n H_{2n+2} + nH_2O$$
 (2.8)

$$nCO + 2nH_2 \leftrightarrow C_n H_{2n} + nH_2 O \tag{2.9}$$

Fischer-Tropsch processes have been used since the 1920s, notably in Germany during WWII and also in South Africa during the Apartheid era, although reactor design and catalyst choice have been refined in recent years [57]. Iron and cobalt are typically selected as catalysts for current commercial FT operations [57].

Product yield and composition vary depending on the process conditions. The yields used in this study are taken from the literature. Petrol and diesel production via gasification and FT synthesis is assumed 5.17% wt and 7.79% wt of the initial biomass feed, respectively [49].

#### 2.3.3 Water gas shift reactions

Syngas from gasification can undergo water gas shift reactions to increase the concentration of hydrogen present within the syngas. The water gas shift combines carbon monoxide and water to form hydrogen and carbon dioxide. It is usually performed in two steps - a high temperature water gas shift (300 - 450 °C) and a low temperature water gas shift (200 °C) [58]. Once the syngas has been subject to water gas shift reactions, it is purified to yield hydrogen gas. Typical yields of hydrogen from gasification and water gas shift reactions are between 70 - 80 kg<sub>H2</sub> per tonne of oven-dry biomass feed [48].

#### 2.3.4 Upgrading bio-oil

Bio-oil produced by fast pyrolysis can be upgraded to produce petrol and diesel. This process consists of three steps: hydrotreating, hydrocracking and distillation [16]. Hydrotreating removes oxygen from the compounds in the bio-oil mixture, which



Figure 2-5 Bio-oil upgrading processes and hydrogen source options.

is followed by hydrocracking long chain hydrocarbons to shorter chain molecules. The products are distilled into diesel and petrol fractions (Figure 2-5).

Hydrogen is required for the hydrotreating and hydrocracking stages. Hydrogen may be delivered to an upgrading facility from an external source, or it may be produced on-site by steam reformation of natural gas or by steam reforming a fraction of the initial bio-oil feed (Section 2.3.5). Upgrading bio-oil to petrol and diesel has only been demonstrated at the laboratory or small engineering development scale, but a recent study performed a techno-economic analysis of a commercial scale facility [16]. Yields of petrol and diesel from upgrading are approximately 31.35% wt and 1.65% wt of the bio-oil feed, respectively [16].

#### 2.3.5 Steam reformation of bio-oil

Approximately 60 - 80% wt of bio-oil is soluble in water [17]. This aqueous fraction can be steam reformed to produce hydrogen in a similar process used to steam reform methane [18]. However, reforming the aqueous fraction of bio-oil is presented with various challenges, most of which are related to coke formation on the surface of catalysts. Adding a solvent, such as methanol, can alleviate these issues, and typically a 10% wt methanol blend is prepared before reformation [19].

Steam reformation of bio-oil has only been demonstrated at laboratory scales (e.g. [59]–[61]), although a recent study provided a techno-economic model of a large scale facility [19]. The maximum stoichiometric yield of hydrogen that can be produced from

steam reformation of bio-oil is 17.2% wt [62]. Steam reformation of a bio-oil methanol blend (10%) yields approximately 14.7% wt hydrogen [19].

#### 2.4 Summary

This Chapter has provided background information on forest residues, fast pyrolysis and torrefaction processes and products, and bio-fuel production processes.

Forest residues are the tree tops and branches from roundwood logging operations. The quantity and availability of a forest residue resource depends on biophysical factors, tree species, logging industry practices, residue recovery methods, residue retention regulations, and competition between industries for access to forest residues.

Fast pyrolysis produces syngas, bio-oil and bio-char. The bio-oil may be used directly as a feedstock for bio-fuel production, or it may be combined with bio-char to produce a bio-slurry that contains a larger fraction of the initial energy content of a forest residue resource. Torrefaction produces syngas and torrefied wood. Bio-oil, bio-slurry and torrefied wood can be used as a feedstock for bio-fuel production processes.

Woodchip, bio-oil, bio-slurry and torrefied wood are all suitable as feedstock for gasification, which produces syngas that can undergo Fischer-Tropsch or water gas shift reactions to produce petrol and diesel, or hydrogen fuels respectively. The bio-oil product of fast pyrolysis may also be upgraded to produce petrol and diesel or steam reformed to produce hydrogen.

The following Chapter introduces the model constructed to investigate the technical and economic impacts of (i) harvesting a forest residue resource using a network of mobile facilities, and (ii) producing bio-fuels using the products of mobile facilities as feedstock.

# 3 Methods

This chapter presents a model for the production of bio-fuels from a forest residue resource. The model contains three main sections: definition of a forest residue resource (Section 3.1), collection and delivery of a forest residue resource to a bio-fuel facility plant gate (Section 3.2), and production of bio-fuels at a bio-fuel facility (Section 3.3). The model tracks both mass and energy flows of the forest residue resource through to bio-fuel production.

Four pathways of delivering a forest residue resource to a bio-fuel facility plant gate are considered: (i) woodchips (ii) bio-oil (iii) bio-slurry and (iv) torrefied wood. Furthermore, two point-of-delivery scenarios are included in the model to account for situations when a bio-fuel facility is either located within or at a distance from the forested region (Section 3.2.1). Four bio-fuel production processes are considered: (i) gasification and Fischer-Tropsch synthesis (ii) gasification and water gas shift reactions (iii) upgrading of bio-oil and (iv) steam reformation of bio-oil. Thus, the model incorporates the entire system of bio-fuel production from collection of forest residues to the synthesis of petrol and diesel, or hydrogen products. The main outputs of the model are a levelised delivered cost of forest residue resource to bio-fuel facility, and a final levelised cost of bio-fuel production. Levelisation converts a series of varying payments into a financially equivalent annuity.

*Products* of mobile facilities and *commodities* are two terms used throughout the description of the model and also in later Chapters. *Products* of mobile facilities are gross quantities of products from mobile facility processes. *Commodities* are net quantities of mobile facility products that can be used as feedstock for bio-fuel production.

Note: All lower case symbols in Equations represent values or data input into the model and all upper case symbols represent values or data calculated by the model.

#### 3.1 Defining a forest residue resource

The model is based upon the amount of a forest residue resource that is available for bio-fuel production. An annual supply of available forest residues (m<sup>3</sup>), *h*, is input by the user. The moisture content of the forest residues, *z*, and the density of oven-dry wood,  $\rho_{ODW}$ , allow the model to calculate the annual forest residue resource in terms of green tonnes (GT),  $M_{GT}$ , and oven-dry tonnes (ODT),  $M_{ODT}$ , using Equations 3.1 and 3.2:

$$M_{GT} = \frac{h}{10^3} \left( \frac{z - z_{ODW}}{\rho_{H_2O}} + \frac{(1 - z) + z_{ODW}}{\rho_{ODW}} \right)^{-1}$$
(3.1)

$$M_{ODT} = M_{GT} (1 - z + z_{ODW})$$
(3.2)

where  $\rho_{H_2O}$  is the density of water and  $z_{ODW}$  is the moisture content of oven-dry wood. The annual energy content,  $E_{FR}$ , within the forest residue resource is then calculated using Equation 3.3:

$$E_{FR} = M_{ODT} l_{ODW} \tag{3.3}$$

where  $l_{ODW}$  is the lower heating value of oven-dry wood.

The extent of the forest residue harvest region is determined by both the volume of forest residues to be collected during the harvest operation lifetime and the spatial density of the available residues,  $\varphi$ . The spatial density of forest residues is assumed to be constant across the land surface, and the harvest region is assumed to be circular (e.g. [63], [64]). Thus, the radius of the harvest region, *R*, is calculated using Equation 3.4:

$$R = \sqrt{hy/\varphi\pi} \tag{3.4}$$

where *y* is the lifetime of the harvest operation.

#### 3.2 Feedstock collection and delivery to bio-fuel facility

Once the forest residue resource has been defined, the model contains four pathways of delivering the resource to a bio-fuel plant gate (Figure 3-1). The point-of-delivery can be selected as a bio-fuel facility either within or outside of the forest residue harvest region (Figure 3-2), as discussed in Section 3.2.1. The model for harvesting and transporting the forest residue resource to a bio-fuel facility as woodchips, bio-oil, bio-slurry or torrefied wood is presented in Section 3.2.2. Technical mobile facility calculations are provided in Section 3.2.3, and all financial calculations related to the delivery of resource to a bio-fuel plant gate are provided in Section 3.2.4.



Figure 3-1 Four pathways for delivering a forest residue resource.



Figure 3-2 Point-of-delivery scenarios.

#### 3.2.1 Point-of-delivery scenarios

Two point-of-delivery scenarios are considered in this study (Figure 3-2). In scenario A, the forest residue resource is delivered to a bio-fuel facility located at the centre of a biomass harvest region. This scenario models a situation where a bio-fuel facility is located within the forested region to purposefully use the local resource. In scenario B, feedstock is delivered to a bio-fuel facility outside of the harvest region, requiring greater transport distances. This scenario models a situation where a bio-fuel facility located at a settlement outside the forested region is the point-of-delivery.

#### 3.2.2 Harvest and transport

Harvest and transport models are important components of determining the cost of delivering a biomass resource to a bio-fuel facility. Models in the literature range from simple continuous models applicable to idealised situations (e.g. [65]), to those incorporating geographical information systems (GIS) that require details of landscape attributes and road networks (e.g. [66]). This study implements a discrete transport model (with no specific geographical setting) similar to those used in the literature (e.g. [7], [63]), with modifications to account for the use of mobile facilities. The harvest and transport model is described in relation to woodchip delivery in Section 3.2.2.1, and pathways involving the use of mobile facilities in Section 3.2.2.2.

#### 3.2.2.1 Woodchip delivery

The conventional method for utilising forest residues as a biomass feedstock requires transporting woodchips directly from the logging fields to a bio-fuel facility. In this process, residues are piled at the roadside where they are chipped directly into a chip truck, which delivers the woodchips to their destination.

The model does not include an in-depth representation of piling and chipping forest residues, such as evaluating machine hours and system productivity, as these vary depending on the equipment used [21]. Therefore the model considers only the cost of piling and chipping forest residues as discussed in Section 3.2.4.1.

The transport model is based upon the assumption that the harvest region is circular (Figure 3-3a). The average annual transport distance from logging fields to a


Figure 3-3 a) Harvest and transport model for delivering a conventional woodchip feedstock.b) Overview of harvest grid for harvest and transport model when implementing mobile facilities.c) A distributed collection region with mobile facility at the centre. d) Average transport distances (scenario A).

bio-fuel facility,  $\overline{D}_{F,B}$ , is calculated using Equation 3.5 (scenario A) or Equation 3.6 (scenario B), depending on the point-of-delivery to a bio-fuel facility:

scenario A 
$$\overline{D}_{F,B} = \tau (2R/3)$$
 (3.5)

scenario B 
$$\overline{D}_{F,B} = \tau ((2R/3) + d)$$
 (3.6)

where  $\tau$  is a tortuosity factor to account for the winding of roads (usually ranging between 1.2 and 3 [63]), and *d* is an additional transport distance to account for the point-of-delivery being located at a distance from the harvest region. The average annual distance woodchips are transported,  $\overline{D}_{WC}$ , is equal to the average annual transport distance from logging fields to bio-fuel facility, i.e.:

$$\overline{D}_{WC} = \overline{D}_{F,B} \tag{3.7}$$

The total annual transport distance travelled by woodchip delivery trucks will depend on whether the truck load is limited by weight or volume of woodchips (Appendix A), and is calculated using Equation 3.14 (Section 3.2.2.2).

## 3.2.2.2 Bio-oil, bio-slurry or torrefied wood delivery

Implementing distributed mobile facilities results in either bio-oil, bio-slurry or torrefied wood delivered to the bio-fuel facility, depending on the mobile facility process selected. Forest residues are chipped into chip trucks and transported to the nearest mobile facility. Preparation of the woodchip feed is followed by either fast pyrolysis or torrefaction, and the associated commodity that is produced is transported to the bio-fuel facility.

The transport model used when implementing mobile facilities is shown in Figure 3-3b and Figure 3-3c. The size of the entire harvest region is defined using Equation 3.4. Smaller square harvest regions are then assumed for distributed collection of forest residues. The number of distributed collection regions depends on the number of mobile facilities required to process the annual harvest, and also the number of times each facility is relocated over the lifetime of the harvest operation. The layout of the

distributed collection regions is assumed such that no overlap occurs, thus representing a gridded version of the entire harvest region. The number of distributed collection regions,  $N_{arid}$ , is calculated using Equation 3.8:

$$N_{grid} = N_{mob}(1+n_r) \tag{3.8}$$

where  $N_{mob}$  is the number of mobile facilities required to meet an annual harvest and  $n_r$  is the number of times each mobile facility is relocated over the lifetime of the harvest operation. The number of mobile facilities required varies depending on the annual harvest because the size of the mobile facilities is fixed. A larger annual harvest will demand more mobile facilities. The number of mobile facilities required,  $N_{mob}$ , is calculated using Equation 3.9:

$$N_{mob} = \left(\frac{M_{ODT}}{s_{mob}\sigma_{mob}}\right) \div 365 \tag{3.9}$$

where  $s_{mob}$  is the size of the mobile facilities (50 ODTPD) and  $\sigma_{mob}$  is the maximum capacity factor of the mobile facilities.

The grid details all the distributed collection regions that will be harvested over the operation lifetime. Each distributed collection region is only occupied once during the operation lifetime.

When harvesting forest residues within a distributed collection region, the average distance from any point in the square collection region to the mobile facility at the centre,  $\overline{D}_{F,M}$ , is calculated using Equation 3.10:

$$\overline{D}_{F,M} = \frac{1}{6} \tau \sqrt{H_{mob}/\varphi} \left(\sqrt{2} + \ln\left(1 + \sqrt{2}\right)\right)$$
(3.10)

where  $H_{mob}$  is the amount of available forest residues within the distributed collection region to be harvested whilst the mobile facility is at that location, defined using Equation 3.11:

$$H_{mob} = \frac{Total \ harvest \ over \ operation \ liftime}{Number \ of \ mobile \ locations} = \frac{hy}{N_{grid}}$$
(3.11)

When mobile facilities are implemented, the average annual delivery distances of woodchips,  $\overline{D}_{WC}$ , or mobile facility commodity j,  $\overline{D}_j$ , are shown by Equations 3.12 and 3.13:

$$\overline{D}_{WC} = \overline{D}_{F,M} \tag{3.12}$$

$$\overline{D}_j = \overline{D}_{M,B} \approx \overline{D}_{F,B} \tag{3.13}$$

where  $\overline{D}_{M,B}$  is the average annual transport distance between mobile facilities and the bio-fuel facility (assumed to be equal to the average annual distance from logging fields to the bio-fuel facility,  $\overline{D}_{F,B}$ , due to the uniform grid of mobile facility locations).

The commodities produced by fast pyrolysis or torrefaction are transported using B-train trucks, assuming mobile facilities are located on sites with road networks adequate for larger vehicles. Once the forest residue resource in one grid location has been depleted, the mobile facility is relocated to an un-harvested grid location.

The total annual transport distance,  $D_j$ , of each commodity, j, (woodchips, bio-oil, bio-slurry or torrefied wood) is calculated using Equation 3.14:

$$D_j = 2\overline{D}_j \frac{M_{j,net}}{M_{truck,j}}$$
(3.14)

where  $M_{j,net}$  is the annual mass of commodity *j* to be transported (Section 3.2.3.4), and  $M_{truck,j}$  is the actual load carried by a truck, which depends on whether trucks are limited by weight or volume (Appendix A).

# 3.2.3 Mobile facilities

Mobile facility product yields are calculated in terms of mass and energy (Section 3.2.3.1). Some of the products of mobile facility processes are used for drying of

woodchips (Section 3.2.3.2) and electricity generation (Section 3.2.3.3), and these requirements are calculated on an energy basis. The net amount of products available as a commodity feedstock for bio-fuel production are discussed in Section 3.2.3.4. Storage requirements at mobile facilities, to maintain a continuous production process, are addressed in Section 3.2.3.5.

#### 3.2.3.1 Mobile facility gross product yields

The annual gross quantity of product *i* from mobile facilities (syngas, bio-oil, bio-char or torrefied wood) is calculated in terms of mass,  $M_{i,gross}$ , and energy,  $E_{i,gross}$ , using Equations 3.15 and 3.16:

$$M_{i,gross} = x_i M_{ODT} \tag{3.15}$$

$$E_{i,gross} = y_i E_{FR} \tag{3.16}$$

where  $x_i$  and  $y_i$  are the mass and energy product ratios provided in Figure 2-2 and Figure 2-3 (Pages 7 and 10). Net quantities of commodities available for bio-fuel production are discussed later in Section 3.2.3.4.

## 3.2.3.2 Drying

For both fast pyrolysis and torrefaction, the produced syngas is used to dry woodchips at the mobile facility. The amount of energy required for drying is calculated based on the initial moisture content of biomass and the efficiency of the dryer [67]. The annual amount of water removed from woodchips,  $M_{H_2O}$ , is calculated using Equation 3.17:

$$M_{H_2O} = M_{GT} - M_{ODT} (3.17)$$

The annual energy required to dry woodchips,  $E_{dry}$ , is calculated using Equation 3.18:

$$E_{dry} = \frac{e_{dry} M_{H_2O}}{\eta_{dry}} \tag{3.18}$$

where  $e_{dry}$  is the energy required to evaporate one tonne of water from woodchips (which is based on the energy required to raise the moisture in the wood from 25°C to 100°C and the latent heat of water at 100°C) and  $\eta_{dry}$  is the drying efficiency [68].

If the annual energy content of produced syngas is not sufficient for the drying of woodchips, the annual amount of propane imported to mobile facilities,  $M_{C_3H_8}$ , is calculated using Equation 3.19:

$$M_{C_3H_8} = \frac{E_{dry} - E_{S,gross}}{l_{C_3H_8}}$$
(3.19)

where  $l_{C_3H_8}$  is the lower heating value of propane.

# 3.2.3.3 Electrical demands

Mobile fast pyrolysis facilities generate electricity using a portion of the produced bio-oil. The fraction of bio-oil product used for electricity generation,  $X_{BO}$ , is calculated using Equation 3.20:

$$X_{BO} = \frac{(\beta M_{ODT})/\eta_{gen}}{E_{BO,gross}}$$
(3.20)

where  $\beta$  is the electricity demand per oven-dry tonne of woodchip feed and  $\eta_{gen}$  is the efficiency of the generator.

Mobile torrefaction facilities are powered by a diesel generator. The electrical demand of a mobile torrefaction facility is assumed the same as a mobile fast pyrolysis facility, although it is likely to be lower as no grinding of woodchips or condensation unit is required. The diesel generator is sized to meet electrical requirements and is fueled from on-site diesel storage available at logging sites. The annual amount of diesel consumed at each mobile facility is calculated on a linear scale using consumption rates

of a 200kW generator (Appendix A) [69]. The total diesel consumption of all mobile facilities is then calculated.

# 3.2.3.4 Mobile facility net product yields (commodities)

The quantities of commodities produced at mobile facilities available as a feedstock for bio-fuel production are calculated as follows. The amount of bio-oil available for bio-fuel production depends on the electrical requirements of mobile facilities. The net quantity of a bio-oil commodity in terms of mass,  $M_{BO,net}$ , and energy,  $E_{BO,net}$ , are calculated using Equations 3.21 and 3.22:

$$M_{B0,net} = M_{B0,gross}(1 - X_{B0})$$
(3.21)

$$E_{BO,net} = E_{BO,gross}(1 - X_{BO})$$
(3.22)

The quantity and properties of a bio-slurry commodity depend on the amount of bio-oil available as well as the bio-char loading of the slurry. The bio-char loading of the bio-slurry, Z, is calculated using Equation 3.23:

$$Z = \frac{M_{BC,gross}}{M_{BO,net} + M_{BC,gross}}$$
(3.23)

However, if the bio-char loading is above the maximum value of 20%wt, the model will reduce the amount of bio-char added to the bio-slurry, and some excess bio-char will remain. The annual amount of excess bio-char,  $M_{BC,excess}$ , is calculated using Equation 3.24:

$$M_{BC,excess} = M_{BC,gross} - \left(\frac{\zeta_{max}M_{BO,net}}{1 - \zeta_{max}}\right)$$
(3.24)

where  $\zeta_{max}$  is the maximum bio-char loading of a bio-slurry (20% wt).

The net annual amount of bio-char added to the bio-slurry is calculated using Equation 3.25:

$$M_{BC,net} = M_{BC,gross} - M_{BC,excess}$$
(3.25)

Therefore, the net annual quantity of bio-slurry delivered to the bio-fuel facility in terms of mass,  $M_{BS,net}$ , and energy,  $E_{BS,net}$ , is calculated using Equations 3.26 and 3.27:

$$M_{BS,net} = M_{BO,net} + M_{BC,net} \tag{3.26}$$

$$E_{BS,net} = E_{BO,net} + \left(\frac{M_{BC,net}}{M_{BC,gross}}\right) E_{BC,gross}$$
(3.27)

The density of the bio-slurry,  $\rho_{BS}$ , is calculated on a linear scale between 1100 kg·m<sup>-3</sup> (0% bio-char loading) and 1300 kg·m<sup>-3</sup> (30% bio-char loading) provided from referenced data [38].

The net quantity of torrefied wood that is delivered to a bio-fuel facility is equal to the gross quantity produced (because torrefied wood is not used to meet any system requirements at mobile torrefaction facilities) as shown by Equations 3.28 and 3.29:

$$M_{TW,net} = M_{TW,gross} \tag{3.28}$$

$$E_{TW,net} = E_{TW,gross} \tag{3.29}$$

#### 3.2.3.5 Storage at mobile facilities

To maintain a continuous fast pyrolysis or torrefaction process a constant supply of woodchips is required. Therefore a storage pile of woodchips at the mobile facility site is necessary. Three days worth of storage for woodchips is assumed to allow for holidays or weekends when woodchip transport may not occur [70]. Storage of mobile facility products is also required, and three days storage is assumed. Bio-oil or bio-slurry product is stored in stainless steel tanks [8], and torrefied wood product is stored in piles.

The maximum storage volume required for commodity j at each mobile facility,  $V_{i,mob}$ , is calculated using Equation 3.30:

$$V_{j,mob} = \frac{3}{365} \left( \frac{M_{j,net} \cdot 10^3}{\rho_j N_{mob}} \right)$$
(3.30)

where  $\rho_j$  is the density of commodity *j*.

# 3.2.4 Costs of feedstock collection and delivery

The model includes financial calculations to produce a levelised cost of collecting and delivering the forest residue resource to a bio-fuel plant gate ('levelised delivered cost') using each of the four delivery pathways. The levelised delivered cost is calculated on an energy basis to provide an equal comparison for each delivery pathway.

The total delivered cost of the forest residue resource to a bio-fuel facility plant gate can be divided into three parts: costs of collecting a forest residue resource, costs relating to mobile facilities, and costs of transportation. Collection costs of a forest residue resource relate to purchasing, piling and chipping forest residues (Section 3.2.4.1). Mobile facility costs include purchasing and operating the mobile facilities, as well as storage requirements (Section 3.2.4.2). Transport costs relate to the hauling of woodchips and mobile facility commodities (Section 3.2.4.3).

## 3.2.4.1 Costs of forest residue resource

The costs of collecting a forest residue resource are the same for each delivery pathway regardless of mobile facility implementation. The annual cost of process p (purchasing, piling or chipping forest residues),  $C_p$ , is calculated using Equation 3.31:

$$C_p = c_p M_{GT} \tag{3.31}$$

where  $c_p$  is the unit cost of process p. (The purchase cost of forest residues is a fee for removal of fuels from the logging field [49]).

#### 3.2.4.2 Mobile facility costs

When woodchips are delivered directly to a bio-fuel facility, no mobile facilities are required and mobile facility costs are equal to zero. Implementing mobile facilities incurs costs associated with the purchase and operation of the facilities.

The size of mobile facilities remains fixed regardless of the annual harvest of forest residues. The capital cost incurred for all mobile facilities,  $C_{mob}$ , is calculated using Equation 3.32:

$$C_{cap,mob} = c_{mob} N_{mob} \tag{3.32}$$

where  $c_{mob}$  is the reference cost for a single mobile facility. For mobile torrefaction facilities, the cost of a diesel generator is also added to the capital cost.

The annual operation and maintenance cost for all mobile facilities,  $C_{O\&M,mob}$ , is calculated using Equation 3.33:

$$C_{O\&M,mob} = o_{mob}C_{cap,mob} \tag{3.33}$$

where  $o_{mob}$  is a fixed operation and maintenance percentage of capital expenditure.

The annual cost of purchasing propane to assist with drying at mobile facilities,  $C_{C_3H_8,mob}$ , is calculated using Equation 3.34:

$$C_{C_3H_8,mob} = c_{C_3H_8} \left( \frac{M_{C_3H_8} \cdot 10^3}{\rho_{C_3H_8}} \right)$$
(3.34)

where  $c_{C_3H_8}$  is the dollar per litre cost of propane.

The annual cost of purchasing diesel for electricity generation at mobile torrefaction facilities,  $C_{diesel,mob}$ , is calculated using Equation 3.35:

$$C_{diesel,mob} = c_{diesel} \left( \frac{M_{diesel} \cdot 10^3}{\rho_D} \right)$$
(3.35)

where  $c_{diesel}$  is the doller per litre cost of diesel,  $M_{diesel}$  is the annual consumption of diesel, and  $\rho_D$  is the density of diesel.

The annual labour costs for all mobile facilities,  $C_{lab,mob}$ , is calculated using Equation 3.36:

$$C_{lab,mob} = n_{staff} c_{wage,mob} N_{mob}$$
(3.36)

where  $n_{staff}$  is the number of staff required at each mobile facility and  $c_{wage,mob}$  is the annual wage of each employee.

Storage of woodchips and torrefied wood at mobile facilities is assumed to be free, as the cost of maintaining open air storage is deemed negligible. Storing bio-oil or bio-slurry incurs costs associated with purchasing and maintaining a stainless steel tank at each mobile facility. The capital cost of all stainless steel tanks for commodity j (bio-oil or bio-slurry),  $C_{tank,mob,j}$ , is calculated using Equation 3.37:

$$C_{tank,mob,j} = N_{mob}c_{ref,tank} \left(\frac{V_{j,mob}}{s_{ref,tank}}\right)^{\kappa_{tank}}$$
(3.37)

where  $s_{ref,tank}$  and  $c_{ref,tank}$  are the reference size and cost of a stainless steel tank, respectively, and  $\kappa_{tank}$  is the cost scaling factor.

The annual maintenance costs of stainless steel tanks,  $C_{O\&M,tank,mob}$ , is calculated using Equation 3.38:

$$C_{0\&M,tank,mob} = o_{tank}C_{tank,mob}$$
(3.38)

where  $o_{tank}$  is a fixed operation and maintenance percentage of capital expenditure.

31

The costs of relocating mobile facilities are distributed evenly throughout the operation lifetime of harvesting forest residues (i.e. the time period between relocations is constant). All mobile facilities are assumed to relocate at the same time. The cost of one relocation of all mobile facilities,  $C_{relocate,mob}$ , is calculated using Equation 3.39:

$$C_{relocate,mob} = N_{mob}c_{relocate} \tag{3.39}$$

where  $c_{relocate}$  is the cost of relocating a single mobile facility.

#### 3.2.4.3 Transport costs

For each commodity (i.e. woodchips, bio-oil, bio-slurry or torrefied wood), there is a fixed cost of transportation and a variable hauling cost. The fixed cost is associated with loading and unloading of trucks. The variable hauling cost depends on truck capacities, the amount of each commodity to be transported, and transport distances as provided by the transport model detailed in Section 3.2.2.

The annual fixed costs of transportation for commodity j,  $C_{tran,fix,j}$ , are calculated using Equation 3.40:

$$C_{tran,fix,j} = c_{tran,fix,j} M_{j,net}$$
(3.40)

where  $c_{trans,fix,j}$  is the unit cost of loading and unloading commodity *j*.

The annual variable cost of transportation for commodity j,  $C_{tran,var,j}$ , is calculated using Equation 3.41:

$$C_{tran,var,j} = c_{tran,var,j} D_j \tag{3.41}$$

where  $c_{tran,var,j}$  is the unit hauling cost for commodity *j*.

No road building costs are included in the model as collection and delivery of forest residues is assumed to occur at existing logging sites, which have road access that has been funded by the lumber industry.

#### 3.2.4.4 Levelised delivered cost

For each delivery pathway, all costs spanning the operation lifetime of harvesting and delivering forest residues are brought to present day value, totaled, and annualised. The levelised delivered cost of commodity j,  $LDC_i$ , is calculated using Equation 3.42:

$$LDC_{j} = \frac{CRF \times NPV_{j}}{E_{j,net}}$$
(3.42)

where *CRF* is the capital recovery factor, and  $NPV_j$  is the net present value of all costs expended to deliver commodity *j*.

#### 3.2.5 Summary

Section 3.2 has provided technical and financial calculations that are performed by the model up to the point of delivering either woodchips, bio-oil, bio-slurry or torrefied wood to a bio-facility plant gate. The first main output of the model, a levelised delivered cost of forest residue feedstock, has been described. Section 3.3 continues the description of the model, providing calculations used to simulate the production of bio-fuels from the commodities delivered to the bio-fuel facility.

# 3.3 **Bio-fuel facilities and production**

The model considers bio-fuel production from each of the four commodities (woodchip, bio-oil, bio-slurry and torrefied wood) delivered to the bio-fuel facility plant gate (Figure 3-4). All are suited for gasification, followed by Fischer-Tropsch or water gas shift reactions. Bio-oil may also be used as a feedstock for upgrading or steam reformation processes.

The sizing of bio-fuel facilities and the quantity of bio-fuels produced depend on the amount of commodity feedstock delivered to the bio-fuel facility. The model sizes facilities appropriately and calculates annual bio-fuel production from each bio-fuel production process as outlined in Sections 3.3.1 and 3.3.2, respectively. Financial calculations provide a levelised cost of bio-fuel production (including all aspects of harvesting a forest residue resource), as discussed in Section 3.3.3.



Figure 3-4 Bio-fuel production options.

# 3.3.1 Bio-fuel facilities

## 3.3.1.1 Feedstock handling and storage

Storage requirements at bio-fuel facilities depend on feedstock reliability, the size of the bio-fuel facility, and risk management requirements imposed by project investors and bio-fuel customers [70]. Greater storage is required for larger facilities as, in addition to requiring more feedstock, larger facilities require greater investments and therefore carry more risk, which means their storage capacities are disproportionally larger that smaller facilities to maintain the production process. If feedstock reliability is low, greater storage is required to ensure continuous production processes. Studies in the literature typically consider two to four weeks of feedstock reserve for large bio-energy facilities [70].

The type of storage employed at a bio-fuel facility depends on the feedstock characteristics, available storage space and the capital available for storage facilities. If storage space is limited, or if woodchips arrive with a low moisture content, woodchips may be stored in hoppers, bins, silos or bunkers. This study assumes green woodchips are stored in large open uncovered piles on top of a base of concrete pads, which is significantly less expensive.

Bio-oil and bio-slurry are stored in stainless steel tanks [8]. Methanol is added to bio-oil to prevent aging during longer term storage (10%wt) [19]. Torrefied wood is stored in open uncovered piles similar to green woodchips.

The model calculates two week reserve storage requirements at central facilities. The volume of storage of commodity j at the bio-fuel facility,  $V_{j,bf}$ , is calculated using Equation 3.43:

$$V_{j,bf} = \frac{14}{365} \left( \frac{M_{j,net} \cdot 10^3}{\rho_j} \right)$$
(3.43)

(Storage of bio-oil includes a methanol additive as part of the mixture, detailed in Section 3.3.1.3).

For liquid storage at central facilities, the number of stainless steel tanks required,  $N_{tank}$ , is calculated based upon instalments of 9,400 m<sup>3</sup> capacity stainless steel tanks [8].

#### 3.3.1.2 Pre-treatment

Green woodchips and torrefied wood need to undergo pre-treatment processes before being input to gasification reactors. Green woodchips need to be screened to remove metal, stone and dirt, before being dried and ground down to finer particles [70]. Torrefied wood only needs to be ground to finer particles. The model does not include an in-depth representation of pre-treatment processes, such as evaluating energy requirements of each process, as these vary depending on the equipment used [21]. Therefore the model considers only the cost of pre-treatment processes provided in the literature, as discussed in Section 3.3.3.1.

#### 3.3.1.3 Sizing of bio-fuel facilities

Bio-fuel facilities are sized depending on the amount of commodity feedstock available for bio-fuel production processes. After commodities have been delivered to the bio-fuel facility, the quantity of feedstock for bio-fuel production processes changes during storage or pre-treatment processes. The quantity and properties of bio-oil are altered when methanol is added, and the mass of woodchips is reduced after drying. The mass quantities of commodity *j* used directly as feedstock for bio-fuel production processes,  $M_{j,feed}$ , are outlined in Equations 3.44 to 3.47:

$$M_{WC,feed} = M_{ODT} \tag{3.44}$$

$$M_{B0,feed} = (1 + m_{meth})M_{B0,net}$$
(3.45)

$$M_{BS,feed} = M_{BS,net} \tag{3.46}$$

$$M_{TW,feed} = M_{TW,net} \tag{3.47}$$

where  $m_{meth}$  is the mass ratio of methanol added to bio-oil (10% wt).

The size (tonne per day) of bio-fuel facility k using commodity feedstock j,  $S_{bf,i,k}$ , is calculated using Equation 3.48:

$$S_{bf,j,k} = \left(\frac{M_{j,feed}}{\sigma_{bf,k}}\right) \div 365 \tag{3.48}$$

where  $\sigma_{bf,k}$  is the capacity factor of bio-fuel facility *k*.

# 3.3.2 Bio-fuel production

When bio-fuels are produced via methods using gasification processes, all feeds are assumed to be different forms of oven-dry woodchips (i.e. they would yield the same ultimate analysis results) and so the same product yield ratios, f, are used [13], [54]. This is because no conversion factors from bio-oil, bio-slurry or torrefied wood to bio-fuels via gasification and Fischer-Tropsch or water gas shift reactions have been identified in the literature. Bio-fuel production from bio-oil via upgrading or steam reformation are available in the literature, and are used in the model.

All bio-fuel production ratios are provided in terms of mass. Further calculations in the model provide conversion to litres, gallons and gigajoules of bio-fuel produced and are detailed in Appendix A.

36

Fischer-Tropsch synthesis yields petrol and diesel. The total annual quantity (kg) of petrol produced from commodity feedstock j,  $F_{kg,j,FT,P}$ , is calculated using Equation 3.49:

$$F_{kg,j,FT,P} = f_{FT} f_{r,FT} M_{j,feed} \cdot 10^3$$
(3.49)

where  $f_{FT}$  is the ratio of bio-fuel yield per unit feed and  $f_{r,FT}$  is the petrol-diesel production ratio.

The total annual quantity (kg) of diesel produced from commodity feedstock j,  $F_{kg,j,FT,D}$ , is calculated using Equation 3.50:

$$F_{kg,j,FT,D} = f_{FT} (1 - f_{r,FT}) M_{j,feed} \cdot 10^3$$
(3.50)

## 3.3.2.2 Water gas shift

The water gas shift yields hydrogen as a product. The total annual quantity (kg) of hydrogen produced from commodity feedstock j,  $F_{kg,j,WGS,H}$ , is calculated using Equation 3.51:

$$F_{kg,j,FT,H} = f_{WGS} M_{j,feed} \cdot 10^3$$
(3.51)

where  $f_{WGS}$  is the ratio of hydrogen production per unit feed.

# 3.3.2.3 Bio-oil upgrading

Bio-oil upgrading yields petrol and diesel as products. Hydrogen is required for the process, which may be derived from steam methane reformation (SMR), steam reformation of a fraction of the bio-oil feed, or from an external hydrogen source.

The annual amount of hydrogen required by the upgrading process,  $M_{H_2,UG}$ , is calculated using Equation 3.52:

$$M_{H_2,UG} = \gamma_{H_2} M_{BO,feed} \tag{3.52}$$

where  $\gamma_{H_2}$  is the ratio of hydrogen required per unit bio-oil feed [16], [71], [72].

If SMR is selected as the method for producing hydrogen, the annual amount of methane consumed,  $M_{CH_4}$ , is calculated using Equation 3.53:

$$M_{CH_4} = \frac{M_{H_2,UG}}{f_{SMR}}$$
(3.53)

where  $f_{SMR}$  is the ratio of hydrogen production from natural gas. The annual methane consumption is used to size the SMR facility at the bio-oil upgrading facility.

If a portion of the bio-oil feed is used to produce hydrogen for upgrading, the fraction of bio-oil feed to be steam reformed,  $Y_{BO,SR}$ , is calculated using Equation 3.54:

$$Y_{BO,SR} = \frac{\gamma_{H_2}}{\gamma_{H_2} + m_{WS} f_{SR}}$$
(3.54)

where  $m_{WS}$  is the percentage of bio-oil that is water soluble and  $f_{SR}$  is the ratio of hydrogen production from steam reformation of the aqueous fraction of bio-oil. The annual mass fractions of bio-oil to be either upgraded or steam reformed are then calculated, and two combined facilities (bio-oil upgrading and bio-oil steam reformation) are sized using Equation 3.48.

No production processes or storage capacities are assumed if hydrogen is sourced externally. Instead, the hydrogen is provided at a cost per kilogram (Section 3.3.3.2).

The total annual quantity (kg) of petrol produced from upgrading bio-oil,  $F_{kg,BO,UG,P}$ , is calculated using Equation 3.55:

$$F_{kg,BO,UG,P} = f_{UG} f_{r,UG} M_{BO,feed} \cdot 10^3$$
(3.55)

where  $f_{UG}$  is the ratio of bio-fuel yield per unit feed and  $f_{r,UG}$  is the petrol-diesel production ratio.

The total annual quantity (kg) of diesel produced from upgrading bio-oil,  $F_{kg,BO,UG,D}$ , is calculated using Equation 3.56:

$$F_{kg,BO,UG,D} = f_{FT} (1 - f_{r,FT}) M_{BO,feed} \cdot 10^3$$
(3.56)

#### 3.3.2.4 Bio-oil steam reformation

Steam reformation of bio-oil yields hydrogen as a product. The total quantity (kg) of hydrogen produced from steam reformation of bio-oil,  $F_{kg,BO,SR,H}$ , is calculated using Equation 3.57:

$$F_{kg,BO,FT,H} = f_{SR} m_{WS} M_{BO,feed} \cdot 10^3$$
(3.57)

where  $m_{WS}$  is the mass ratio fraction of bio-oil that is water soluble, as only the aqueous fraction of bio-oil is suitable for steam reformation.

## 3.3.3 Costs of bio-fuel production

The costs of producing bio-fuels discussed in this Section are those related to bio-fuel facilities and bio-fuel production processes only. Costs associated with feedstock collection and delivery to the bio-fuel facility are discussed earlier in Section 3.2.4.

## 3.3.3.1 Storage and pre-treatment

Woodchip and torrefied wood are assumed to be stored in open-air piles [70], [73]. The annual cost of this storage method for commodity j (woodchip or torrefied wood),  $C_{pile,bf,j}$ , is calculated using Equation 3.58:

$$C_{pile,bf,j} = c_{pile} V_j \tag{3.58}$$

where  $c_{pile}$  is the cost per cubic metre to store commodity *j*, and *V<sub>j</sub>* is the annual volume of commodity *j* delivered to the bio-fuel facility.

Bio-oil and bio-slurry are stored in stainless steel tanks. The capital cost of the storage equipment,  $C_{tank,bf}$ , is calculated based on the number of tanks installed,  $N_{tank}$ , as shown in Equation 3.59:

$$C_{tank,bf} = c_{ref,tank} N_{tank} \tag{3.59}$$

where  $c_{ref,tank}$  is the reference cost of a stainless steel tank.

The annual operation and maintenance costs of the stainless steel tanks,  $C_{0\&M,tank,bf}$ , is calculated using Equation 3.60:

$$C_{0\&M,tank,bf} = o_{tank}C_{tank,cent}$$
(3.60)

where  $o_{tank}$  is the fixed operation and maintenance percentage of the capital expenditure.

The annual cost of methanol,  $C_{meth,bf}$ , added to bio-oil is calculated using Equation 3.61:

$$C_{meth,bf} = c_{meth} m_{meth} M_{B0,net} \cdot 10^3 \tag{3.61}$$

where  $c_{meth}$  is the dollar per kilogram price of methanol.

Feed preparation for woodchips consists of screening for metals and other impurities, drying and grinding. Feed preparation for torrefied wood only requires grinding. The costs associated with these processes are considered additional capital costs due to the additional equipment required at the bio-fuel facility (costs are provided in Appendix B) [49], [50]. This equipment is not required for liquid feedstock, and therefore these additional costs are not included when bio-oil or bio-slurry feedstock are used. The pre-treatment capital cost is added to the bio-fuel facility capital cost and scaled as necessary depending on the bio-fuel facility size (Section 3.3.3.2).

#### 3.3.3.2 Capital, operating and maintenance costs

The capital cost of bio-fuel facility k utilising commodity j as feedstock,  $C_{cap,bf,j,k}$ , is calculated using Equation 3.62:

$$C_{cap,bf,j,k} = c_{ref,k} \left(\frac{S_{bf,j,k}}{s_{ref,k}}\right)^{\kappa}$$
(3.62)

where  $c_{ref,k}$  is the cost of a bio-fuel facility of reference size  $s_{ref,k}$  (tonne per day), and  $\kappa$  is the cost scaling factor.

If woodchip or torrefied wood is used as a feedstock, the capital reference cost includes feed preparation equipment. For woodchip feedstock this requires screening, drying and grinding equipment. For torrefied wood feedstock only a grinder is required. These additional costs are added to the reference capital cost of the bio-fuel facility, and are therefore subject to operation and maintenance costs as well [49].

Equation 3.62 is also used to calculate the capital cost of SMR facilities (when hydrogen is produced from methane at a bio-oil upgrading facility), except that annual methane consumption  $(M_{CH_4})$  is used for scaling costs rather than annual feedstock requirements  $(S_{bf})$ .

When upgrading uses hydrogen sourced from bio-oil steam reformation, Equation 3.62 is used to calculate the capital cost of both an upgrading facility and steam reformation facility separately. These costs are combined to produce a total capital cost (no cost savings other than labour are assumed through co-location of facilities).

The fixed operation and maintenance cost of bio-fuel facility k utilising commodity feedstock j,  $C_{O\&M,bf,j,k}$ , is calculated using Equation 3.63:

$$\mathcal{C}_{0\&M,bf,j,k} = o_{bf,k} \mathcal{C}_{cap,bf,j,k} \tag{3.63}$$

where  $o_{bf,k}$  is a fixed operation and maintenance percentage of capital expenditure.

Additional operating costs that are calculated by the model are those related to the purchase of methane or hydrogen when upgrading bio-oil. The annual cost of purchasing methane for steam reformation to hydrogen,  $C_{CH_4,bf}$ , is calculated using Equation 3.64:

$$C_{CH_4,bf} = c_{CH_4} l_{CH_4} M_{CH_4}$$
(3.64)

where  $c_{CH_4}$  is the dollar per GJ cost of methane and  $l_{CH_4}$  is the lower heating value of methane.

The cost of purchasing hydrogen directly from an external source,  $C_{external-H_2,bf}$ , is calculated using Equation 3.65:

$$C_{external-H_2,bf} = c_{external-H_2} M_{H_2,UG} \cdot 10^3$$
(3.65)

where  $c_{external-H_2}$  is the dollar per kilogram cost of hydrogen from an external source.

The labour costs for the bio-fuel facilities are based on the size of the bio-fuel facility. For a bio-fuel facility size of 2000 TPD, fifty operating and four administrative staff with an hourly wage of \$40 and \$64, respectively, is assumed. The number of administrative workers is assumed constant for all plant sizes. For every increase or decrease in size of bio-fuel facility by 100 TPD, one more or one less operating worker is required, respectively [49].

## 3.3.3.3 Levelised cost of bio-fuel production

For each bio-fuel production route (Figure 3-4), all costs spanning the operation lifetime of harvesting forest residues and producing bio-fuels are brought to present day value, totaled, and annualised. These costs include those calculated in Section 3.2. The levelised cost of producing bio-fuel m from commodity feedstock j at bio-fuel production facility k in terms of mass (kg),  $LCB_{kg,j,k,m}$ , is calculated using Equation 3.66:

$$LCB_{kg,j,k,m} = \frac{CRF \times NPV_{j,k,m}}{F_{kg,j,k,m}}$$
(3.66)

where *CRF* is the capital recovery factor,  $NPV_{j,k,m}$  is the net present value of all costs expended to deliver commodity *j* to bio-fuel facility *k* and produce bio-fuel *m*, and  $F_{kg,j,k,m}$  is the total quantity (kg) of bio-fuel *m* produced.

The levelised cost of producing bio-fuels can be calculated in terms of mass (Equation 3.66), volume or energy content of the final fuels. To provide comparison between bio-fuel production routes in later Chapters, the levelised cost of bio-fuel production may be referred to in dollars per GJ, dollars per litre, or dollars per kilogram of fuel.

#### 3.3.4 Summary

Section 3.3 has provided technical and financial calculations that model the production of bio-fuels from woodchip, bio-oil, bio-slurry or torrefied wood feedstock. The second main output of the model, a levelised cost of bio-fuel production, has been described. Section 3.4 describes the analyses that are performed using the model described in Section 3.1 to Section 3.3 to investigate bio-fuel production when using mobile facilities to deliver a forest residue resource to a bio-fuel facility.

## 3.4 Analysis performed

This Section outlines the analyses that are performed using the model described in Section 3.1 to Section 3.3. All feedstock delivery pathways and bio-fuel production routes are examined. There are four delivery pathways considered, and a total of twelve routes for bio-fuel production - eight bio-fuel production routes via gasification of feedstock, and four bio-fuel production routes specific to a bio-oil feedstock (Figure 3-4). Each time the model is run, results for all four delivery pathways and all twelve routes for bio-fuel production are obtained.

Initially, a base analysis is performed using the model (Section 4.1). For this analysis, an annual harvest of 1.717 million  $m^3$  (equivalent to 2000 ODT) of forest residues is input (which is a standard base annual harvest used by other studies in the

literature) and delivery scenario A is selected. The spatial density of forest residues assumed for the base analysis is 65 m<sup>3</sup> km<sup>-2</sup>, which is consistent with temperate forest data used in other studies (e.g. [49]). The base analysis therefore considers a harvest region of 528,308 km<sup>2</sup> over a 20 year harvest operation. The forest residues are modelled with an initial moisture content of 50%, and an oven-dry wood bulk density and LHV of 500 kg m<sup>-3</sup> and 18 MJ kg<sup>-1</sup>, respectively. Base analysis results are validated against results in the literature and a comparison of delivered feedstock costs, bio-oil production costs, and bio-fuel production costs is provided.

A sensitivity study is then performed to examine the key factors that influence model results (Section 4.2). All input parameters are varied individually by plus and minus 50% of their original value, and the impact on levelised delivered cost of feedstock (LDC) and levelised cost of bio-fuels (LCB) are recorded. Parameters that influence both the LDC and LCB by less than 10% are not subject to further analysis.

Using the results of the sensitivity study, key parameters (those that produce more than 10% variation in LDC or LCB results) are used to identify topics for further analysis. Other issues relating to forest residue resource volumes and characteristics, mobile facilities, and point-of-delivery scenarios are also studied, providing extensive investigation into LDC and LCB results. Results specific to levelised delivered costs of feedstock (LDC) are provided in Section 4.3, and results relating to levelised cost of bio-fuels (LCB) are provided in Section 4.4.

## 3.5 Summary

Chapter 3 has provided a description of the model constructed to investigate the technical and economic impact of producing bio-fuels when using mobile facilities to deliver a forest residue resource to a bio-fuel facility.

The model contains three main sections: definition of a forest residue resource, collection and delivery of a forest residue resource to a bio-fuel facility plant gate, and production of bio-fuels at a bio-fuel facility. The two main outputs of the model are a levelised delivered cost of forest residue resource to a bio-fuel facility plant gate, and a final levelised cost of bio-fuel production. The analyses performed using the model have been described and the results of these analyses are provided next in Chapter 4.

# 4 Results and Discussion

This Chapter provides results and discussion for analyses performed using the model outlined in Chapter 3, and is divided into four Sections. In Section 4.1, base analysis results are presented and validated against existing results in the literature. A sensitivity study is then performed in Section 4.2 to determine key parameters of the model that influence results. A study of feedstock collection and delivery to bio-fuel facilities is presented in Section 4.3, examining in depth each of the four delivery pathways used to deliver a forest residue resource to a bio-fuel facility (Figure 3-1). Finally, Section 4.4 provides a detailed analysis of bio-fuel production costs via each of the bio-fuel production facilities considered in this study (Figure 3-4), using feedstock from each of the four delivery pathways.

Throughout this Chapter, costs of individual processes (from harvesting of forest residues up to bio-fuel production) may be grouped into the categories shown in Table 4-1, for ease of discussion.

biomass	all costs related to the purchase, piling and chipping of forest residues
mobile	all costs related to the purchase, operation and labour requirements of
	mobile facilities, including on-site storage
transport	all costs related to the transport of woodchips and mobile commodities
central	all costs related to the purchase, operation and labour requirements of the
	bio-fuel facility, including on-site storage
hydrogen	all costs related to the production or purchase of hydrogen necessary for the
	bio-oil upgrading process

# Table 4-1 Cost categories.

## 4.1 Base analysis

The base analysis, described in Section 3.4, uses input values provided in Appendix B. Results of the base analysis are discussed in terms of levelised delivered cost of feedstock (Section 4.1.1) and levelised cost of bio-fuel (Section 4.1.2), before validating model results against results available in the literature (Section 4.1.3).

## 4.1.1 Levelised delivered cost

Results for levelised delivered cost of feedstock (LDC) for the base analysis are shown in Table 4-2. The LDC of woodchips is 7.97  $GJ^{-1}$  while the LDC for bio-oil, bio-slurry and torrefied wood are 18.20  $GJ^{-1}$ , 13.03  $GJ^{-1}$  and 8.58  $GJ^{-1}$ , respectively. The annual average transport distance is 273 km. Approximately 80% of the total LDC for woodchip delivery is due to transport costs. Implementing mobile facilities reduces the impact of transport costs to less than 40% of the total LDC for mobile facilities outweigh the reduction in transport costs.

Delivering woodchips ensures that all of the energy content of forest residues are delivered to the bio-fuel facility. When delivering bio-oil, bio-slurry or torrefied wood the total energy delivered to the bio-fuel facility is reduced due to the energy requirements of the mobile facilities, which use a fraction of the biomass energy content to meet thermal or electrical demands. Fast pyrolysis facilities use 12.4% of the produced bio-oil for electricity generation for the facility. Furthermore, the energy content of the char product of fast pyrolysis is not available for bio-fuel production when only bio-oil is delivered. Some of the char is transported when a bio-slurry is produced, although the configuration of the fast pyrolysis reactor in this study yields excess char that cannot be added to the bio-slurry due to a maximum char loading of 20%. Therefore, the total energy delivered by bio-oil, bio-slurry and torrefied wood is 45%, 65% and 87% of the initial forest residue energy content, respectively. This reduced quantity of energy delivered to the bio-fuel facility increases the LDC of each commodity.

Cost component (Table 4-1)	woodchip		bio-oil		bio-slurry		torrefaction	
	\$ GJ <sup>-1</sup>	% <sup>a</sup>	\$ GJ <sup>-1</sup>	% <sup>a</sup>	\$ GJ <sup>-1</sup>	% <sup>a</sup>	\$ GJ <sup>-1</sup>	% <sup>a</sup>
biomass								
purchase, piling & chipping	1.67	20.9	3.70	20.3	2.56	19.6	1.92	22.4
mobile								
feed preparation	-	-	3.74	20.6	2.58	19.8	0.00	0.0
purchase & maintenance	-	-	3.83	21.0	2.65	20.3	2.19	25.5
labour	-	-	2.71	14.9	1.88	14.4	1.41	16.4
relocation	-	-	0.01	0.0	0.00	0.0	0.00	0.0
storage	-	-	0.04	0.2	0.03	0.2	0.00	0.0
transport								
woodchip delivery	6.30	79.1	1.61	8.8	1.11	8.5	0.84	9.8
product delivery	-	-	2.56	14.1	2.22	17.0	2.22	25.9
total	7.97	100.0	18.20	100.0 <sup>b</sup>	13.03	100.0 <sup>b</sup>	8.58	100.0

 Table 4-2 Breakdown of levelised delivered cost (LDC) for base harvest.

<sup>a</sup> Percentage of total LDC

<sup>b</sup> Rounded values do not sum to 100.0%

# 4.1.2 Levelised cost of bio-fuel

Results for levelised cost of bio-fuel (LCB) for the base analysis are shown in Figure 4-1. The net present value of all costs expended for each bio-fuel production route is similar, between \$1.0 - 1.3 billion (Figure 4-1a). However, when the quantity or energy content of bio-fuels produced from each process is taken into account, the results for LCB vary significantly (Figure 4-1b).

Bio-fuels produced via gasification and Fischer-Tropsch synthesis (petrol and diesel) are most expensive, for any feedstock. The variation between Fischer-Tropsch (FT) bio-fuel production costs using different feedstock is due to the quantity of each feedstock delivered to the bio-fuel facility for a given harvest (as mentioned in Section 4.1.1). The LCB from FT synthesis is 41.75 \$ GJ<sup>-1</sup> (1.42 \$ litre<sup>-1</sup>), 66.76 \$ GJ<sup>-1</sup> (2.28 \$ litre<sup>-1</sup>), 60.54 \$ GJ<sup>-1</sup> (2.06 \$ litre<sup>-1</sup>) and 52.92 \$ GJ<sup>-1</sup> (1.80 \$ litre<sup>-1</sup>) for woodchip, bio-oil, bio-slurry and torrefied wood feedstock, respectively.

Upgrading bio-oil to produce petrol and diesel yields a lower LCB than FT bio-fuels, regardless of the source of hydrogen required for upgrading. Upgrading



**Figure 4-1** a) Net present value of costs for all bio-fuel production routes. b) Levelised cost of bio-fuel for all bio-fuel production routes. [Base harvest (1.717 million  $m^3$ ); SMR: Steam methane reformation; BO SR: Bio-oil steam reformation; P: Purchased hydrogen (\$1.5 kg<sup>-1</sup>)].

bio-oil using hydrogen produced by steam reforming a fraction of the bio-oil feed requires less overall expenditures (Figure 4-1a), but less bio-fuel is produced resulting in a higher LCB compared to other hydrogen sources (Figure 4-1b). The LCB from bio-oil upgrading is 27.48 GJ<sup>-1</sup> (0.88 litre<sup>-1</sup>), 35.46 GJ<sup>-1</sup> (1.14 litre<sup>-1</sup>) and 27.46 GJ<sup>-1</sup> (0.88 litre<sup>-1</sup>) when using hydrogen sourced from SMR, bio-oil steam reformation and purchased hydrogen (1.5 kg<sup>-1</sup>), respectively.

Hydrogen produced via gasification and water gas shift (WGS) of woodchip feedstock provides the lowest LCB of all bio-fuel production routes (Figure 4-1b). Hydrogen production via gasification and WGS for other feedstock types is more expensive due to reduced quantities of feedstock available from mobile facilities for bio-fuel production. The LCB from gasification and WGS is 25.06 \$ GJ<sup>-1</sup> (3.01 \$ kg<sup>-1</sup>), 40.21 \$ GJ<sup>-1</sup> (4.83 \$ kg<sup>-1</sup>), 36.45 \$ GJ<sup>-1</sup> (4.37 \$ kg<sup>-1</sup>) and 31.81 \$ GJ<sup>-1</sup> (3.82 \$ kg<sup>-1</sup>) for woodchip, bio-oil, bio-slurry and torrefied wood feedstock, respectively.

Hydrogen production via steam reformation of bio-oil has a levelised cost of  $27.42 \text{ GJ}^{-1} (3.29 \text{ kg}^{-1})$ .

## 4.1.3 Validation

The model presented in Chapter 3 consists of a series of components and processes, each of which is based on methods used by other studies in the literature. Together, these components and processes model the delivery of a forest residue resource to a bio-fuel facility using mobile facilities, as well as bio-fuel production. Validating outputs of the model ensures that combining model components and processes still yields reasonable results.

Table 4-3 provides a comparison of model results to those available in the literature. None of the results in the literature are specifically related to bio-fuel production from forest residues when using mobile facilities. Therefore, an absolute comparison cannot be made, although it is useful to show that the model does provide results within a reasonable range suggested by other studies.

Subject of validation	Model	Literature	Reference(s)
Delivered woodchip feedstock cost ( $\cdot GT^{-1}$ )	70.27	15 - 75 <sup>a</sup>	[35], [48]
Delivered bio-oil feedstock cost (\$·litre <sup>-1</sup> )	0.33	0.30 - 0.58 <sup>b</sup>	[10], [19]
Delivered bio-slurry feedstock cost (\$·litre <sup>-1</sup> )	0.30	0.12 - 0.15 <sup>c</sup>	[39]
Bio-oil production cost using mobile pyrolysis facilities		0.25 <sup>d</sup>	[74]
(\$·litre <sup>-1</sup> )			
Torrefied wood production cost (\$.tonne <sup>-1</sup> )	196.85	68.07 - 95.29 <sup>e</sup>	[13], [43]
Woodchip gasification and Fischer-Tropsch: petrol	1.42	0.76 - 2.40	[49]
and diesel fuel production cost (\$·litre <sup>-1</sup> )			
Bio-oil gasification and Fischer-Tropsch: petrol and	2.28	0.40 - 0.89 <sup>f</sup>	[75], [76]
diesel fuel production cost (\$·litre <sup>-1</sup> )			
Bio-slurry gasification and Fischer-Tropsch: petrol	2.06	1.73 <sup>g</sup>	[56]
and diesel fuel production cost (\$·litre <sup>-1</sup> )			
Woodchip gasification and water gas shift: hydrogen fuel	3.01	2.80 - 5.40	[48]
production cost $(\$ \cdot kg_{H^2}^{-1})$			
Bio-oil upgrading: petrol and diesel fuel			
production cost (\$·litre <sup>-1</sup> )			
Hydrogen sourced from SMR	0.88	0.48 - 0.92	[16], [74]
Hydrogen sourced from bio-oil steam reformation	1.14	1.80	[71]
Purchased hydrogen ( $\$1.5 \text{ kg}_{\text{H2}}^{-1}$ )	0.88	0.90	[71]
Bio-oil steam reformation: hydrogen fuel production cost	3.29	3.12 <sup>b</sup>	[19]
$(\$ kg_{H2}^{-1})$			

 Table 4-3 Validation of the model against results available in the literature.

Model results correspond to an annual harvest of 1.717 million m<sup>3</sup> (2000 ODTPD equivalent) of forest residues. <sup>a</sup> values provided in ODT are converted to GT for comparison using conversion factors in Table 2-1. <sup>b</sup> values are for bio-oil produced from larger scale fixed distributed fast pyrolysis facilities <sup>c</sup> values are for bio-slurry produced from larger scale (500 ODTPD) fixed distributed fast pyrolysis facilities. <sup>d</sup> value is for a larger mobile fast pyrolysis facility (100 ODTPD) <sup>e</sup> values are for larger scale fixed torrefaction facilities, and do not include feedstock costs. <sup>f</sup> values are for larger scale fixed distributed fast pyrolysis facilities providing bio-oil feedstock. <sup>g</sup> values are for bio-slurry sourced from larger scale fixed facilities.

# 4.2 Sensitivity study

A sensitivity study is performed on all input parameters to determine key factors that significantly influence model results. All parameters are varied individually by plus and minus 50% of their original input value, and the impact on levelised delivered cost (LDC) and levelised cost of bio-fuels (LCB) is recorded. Parameters that influence LDC or LCB results by more than 10% are discussed in Sections 4.2.1 and 4.2.2, respectively. Tables detailing the results of the sensitivity study can be found in Appendix C.

## 4.2.1 Levelised delivered cost

Parameters that significantly influence LDC results (produce more than 10% variation in LDC results) can be grouped into the following categories: mobile facility capital cost; densities of commodities; transport factors; mobile facility electrical and drying requirements; and initial moisture content of forest residues.

Mobile facility capital cost significantly influences LDC results as any change in cost is multiplied by the number of installations, and operation and maintenance costs of mobile facilities are also impacted, which are calculated as a percentage of capital cost.

The density of each commodity (woodchip, bio-oil, bio-slurry, or torrefied wood) proves significant as this is used to determine how much of the commodity can be loaded into a single truck (within the weight limit of the vehicle). Thus, the density of a commodity is more important when trucks are limited by volume rather than weight. The LDC of woodchips is particularly sensitive to the density of woodchips.

All transport parameters input to the model, except fixed costs (loading and unloading), are found to be significant by the sensitivity study. Truck volumes impact woodchip and torrefied wood LDC values more than those for bio-oil and bio-slurry delivery. Bio-oil and bio-slurry are impacted by truck weight limits, because liquid loads are limited by weight rather than volume. The tortuosity factor and haul cost factors influence the LDC of woodchip and torrefied wood delivery require greater total transport distances.

The efficiency of the electricity generator at mobile fast pyrolysis facilities is a significant parameter for fast pyrolysis pathways because this affects how much bio-oil is

consumed by the electricity generator and thus the net quantity of bio-oil available as a commodity or for bio-slurry production. The woodchip dryer efficiency is significant for all mobile facilities as this determines how much syngas is used for the drying of woodchips. When the dryer has a low efficiency, propane requirements increase if there is not sufficient syngas for drying, which increases LDC results. The cost of propane also proves important for mobile fast pyrolysis facilities (which produce less syngas than mobile torrefaction facilities).

The initial moisture content of forest residues is a significant parameter for two reasons. First, increased moisture content of forest residues increases the drying requirements at mobile facilities. Second, moisture content determines how much wood content there is within a volume of forest residues i.e. increasing moisture content within a fixed volume of a forest residue resource results in a decreased wood content (Equation 3.1). Therefore, because a volume of forest residues is input to the model, the total energy content of the forest residue resource is less when the moisture content is increased, resulting in higher LDC values (Equation 3.42).

## 4.2.2 Levelised cost of bio-fuel

Parameters that significantly influence LCB results (produce more than 10% variation in LCB results) can be grouped into the following categories: density of woodchips; density of torrefied wood; transport factors; mobile facility electrical and drying requirements; initial moisture content of forest residues; bio-fuel facility capital cost; discount factor; and bio-fuel production factors.

The densities of woodchip and torrefied wood, drying and electrical requirements of mobile facilities, and the initial moisture content of forest residues are significant parameters for LCB results for the same reasons explained in Section 4.2.1.

Bio-fuel facility capital cost influences LCB results significantly as it is a large expenditure and also impacts the operation and maintenance costs of bio-fuel facilities, which are calculated as a percentage of capital cost.

The discount rate input to the model, which reflects interest rates, inflation and other financial risk, is used in financial calculations (in particular for calculating the net present value of future costs) and significantly impacts the levelised cost of bio-fuels for all delivery pathways and bio-fuel production routes. The influence of the discount factor is not major for LDC results, but is more apparent when dealing with large costs associated with the operation and maintenance of bio-fuel facilities.

Parameters that influence the quantity of bio-fuel produced from any feedstock influence LCB results most significantly. A reduction in the quantity of bio-fuel produced will result in a higher LCB (Equation 3.66).

# 4.3 Feedstock collection and delivery

This Section presents further results for each pathway used to deliver a forest residue resource to a bio-fuel facility (Figure 3-1). Levelised delivered costs are provided for a range of forest residue harvest volumes and spatial densities (Section 4.3.1 and Section 4.3.2). Initial moisture content of forest residues and commodity transport factors are studied in more detail, given their influence on LDC results as shown by the sensitivity study (Section 4.3.3 and Section 4.3.4). Relocation of mobile facilities is investigated (Section 4.3.5), and results are examined for a range of point-of-delivery distances (scenario B), modelling a situation where the bio-fuel facility is located at a distance from the forested region (Section 4.3.6).

Note: Results shown throughout this Section relate to bio-fuel facilities located at the centre of the forested region (scenario A), except when discussing point-of-delivery distances in Section 4.3.6.

#### 4.3.1 Annual volume of forest residues

Figure 4-2 shows how the levelised delivered cost (LDC) for each delivery pathway varies with annual harvest volume of forest residues. Above a harvest of approximately 2.3 million m<sup>3</sup> per year it becomes more economical to implement mobile torrefaction facilities. The number of mobile facilities required increases with annual harvest volume (the size of the mobile facilities is fixed). However, the cost reduction provided on conventional woodchip transport direct to a bio-fuel facility means implementing mobile torrefaction facilities reduces the total LDC of resource. The



**Figure 4-2** a) Levelised delivered cost over a range of annual harvest volumes. b) Cost components (Table 4-1) for a bio-fuel facility of size 500 ODTPD (woodchip equivalent). c) Cost components for a bio-fuel facility of size 5000 ODTPD (woodchip equivalent). The quantity of commodity delivered to the bio-fuel facility after conversion processes at mobile facilities is less than when delivering woodchips, and so the bio-fuel facility size required for each possible feed is also provided in Figure 4-2b and Figure 4-2c.

minimum LDC of commodities when using a mobile facility network (for both fast pyrolysis or torrefaction) occurs at an annual forest residue harvest of approximately 200,000 m<sup>3</sup>, which requires the use of only five mobile facilities. However, the average transport distance to the bio-fuel facility for this size of harvest is only 93 km, and conventional woodchip delivery is the lowest cost pathway.

Figure 4-2b and 4-2c show LDC results partitioned into three cost components (biomass, mobile, and transport as discussed in Table 4-1) for annual harvests of 0.429 and 4.292 million m<sup>3</sup> (500 and 5000 ODTPD woodchip equivalent). As the annual harvest increases, a larger portion of the LDC of woodchips is due to transport. The portion of LDC attributed to transport is considerably smaller for pathways using mobile facilities. Therefore, the LDC of resource for mobile facility pathways increases at a lower rate than woodchip delivery as greater volumes of forest residues are harvested.

## 4.3.2 Forest residue spatial density

The spatial density of forest residues affects the LDC of woodchips more than pathways using mobile facilities (Figure 4-3). Varying the spatial density between 30 and 100 m<sup>3</sup> km<sup>-2</sup> results in a +34.2% and -14.1% change in LDC for woodchip delivery, respectively. The LDC of commodities from mobile facilities varies by a maximum and minimum of +12.1% and -5.0% over the same spatial density range. Mobile torrefaction facilities provide the lowest cost delivery pathway when the average spatial density of a forest residue resource is below 47 m<sup>3</sup> km<sup>-2</sup> (for an annual harvest of 1.717 million m<sup>3</sup>).

#### 4.3.3 Initial moisture content

Moisture content of forest residues impacts the feed preparation energy requirements of mobile facilities. Each mobile fast pyrolysis facility requires 1855 litres of propane per day when the initial moisture content is 50%. This quantity of propane accounts for approximately 20% of the LDC for bio-oil or bio-slurry commodities. For the configuration of the mobile facilities assumed in this study, no propane is required to dry woodchips at mobile fast pyrolysis facilities when the initial moisture content of the forest residues is below 37%. Each mobile torrefaction facility produces enough syngas to meet drying requirements when initial moisture content is 50%, and propane is only



Figure 4-3 Levelised delivered costs for a range of forest residue spatial densities.

required to meet drying energy requirements when the moisture content is over 52%. The effect of initial moisture content on LDC is shown in Figure 4-4a.

Propane requirements are also determined by the efficiency of the dryer at mobile facilities. Figure 4-4b shows how propane requirements change when the efficiency of the dryer is 52% (min value), 72% (base value) and 89% (max value) over a range of initial moisture content.

# 4.3.4 Transport

The sensitivity study showed that transport parameters used in the model significantly influenced LDC results. Table 4-4 shows how LDC results vary for each delivery pathway for an annual harvest of 1.717 million m<sup>3</sup> (2000 ODTPD woodchip equivalent) when different haul cost factors are used. Table 4-4 also includes LDC results for when smaller straight trucks are used for mobile facility commodity delivery, which may be necessary when road conditions are not suitable for large B-train vehicles.


Figure 4-4 a) Levelised delivered costs for a range of initial moisture content of forest residues.b) Daily propane requirements at mobile facilities for a range of drying efficiencies and initial moisture content of forest residues (minimum efficiency 52%; maximum efficiency 89%).

	Truck haul cost (\$ km <sup>-1</sup> )					
	1.0	1.5	2.0	2.5	3.0	
Truck type	Total L	evelised	Delivered	d Cost (\$	$GJ^{-1}$ )	
Woodchip						
Straight truck <sup>a</sup>	6.04	7.97	9.89	11.81	13.74	
Bio-oil						
Straight tanker <sup>b</sup>	17.79	18.70	19.61	20.52	21.44	
B-train tanker <sup>c</sup>	16.86	17.31	17.75	18.20	18.64	
Bio-slurry						
Straight tanker <sup>b</sup>	12.68	13.47	14.25	15.04	15.83	
B-train tanker <sup>c</sup>	11.87	12.26	12.64	13.03	13.41	
Torrefied wood						
Straight truck <sup>a</sup>	8.86	9.99	11.11	12.23	13.36	
B-train truck <sup>d</sup>	7.40	7.79	8.19	8.58	8.97	

Table 4-4 Levelised delivered costs for a range of truck types and haul costs.

<sup>a</sup>Straight truck: Volume 70 m<sup>3</sup>, Weight limit 22.5 tonnes [7] <sup>b</sup>Straight tanker: Volume 30 m<sup>3</sup>, Weight limit 30.5 tonnes [77] <sup>c</sup>B-train truck: Volume 200 m<sup>3</sup>, Weight limit 62.5 tonnes [78] <sup>d</sup>B-train tanker: Volume 60 m<sup>3</sup>, Weight limit 62.5 tonnes [79]. Bold values indicate results when base analysis values are used (Section 4.1).

Results show that if the haul cost for woodchip delivery is as low as 1 \$ km<sup>-1</sup>, woodchip delivery will be the lowest cost pathway for delivery of a forest residue resource regardless of other vehicle types or haul cost factors used for other commodity delivery (for the ranges shown in Table 4-4). If the haul cost for woodchip delivery is as high as 3 \$ km<sup>-1</sup>, then torrefied wood is the lowest cost delivery pathway, when either straight trucks or B-train trucks are used. The LDC of bio-slurry may also be less expensive than woodchip delivery (at 3 \$ km<sup>-1</sup>) depending on the vehicle type used and the haul cost of bio-slurry delivery, although torrefied wood remains the lowest cost pathway. Bio-oil is the most expensive delivery pathway, and the LDC of bio-oil is not below that of woodchip delivery, even when the haul cost of woodchips is 3 \$ km<sup>-1</sup>.

However, Table 4-4 (as well as the results of the sensitivity study) demonstrates that the LDC of mobile facility commodities are less influenced by haul cost factors than woodchip delivery. This is because transport requirements are reduced when mobile facility pathways are used for delivery of a forest residue resource. For the same reason, decreasing the size of truck used to transport mobile commodities imparts small variation of LDC results, especially for liquid commodities.

The tortuosity factor used in the model is akin to the nature of the road network (the windiness of roads). This factor was found to be significant in the sensitivity study. Results of varying the tortuosity factor can be inferred from Table 4-4 - total transport costs are calculated by multiplication of both the tortuosity factor and haul cost factor (see Equations in Section 3.2.2). For example, doubling the haul cost factor is analogous to doubling the tortuosity factor, provided the other factor is held constant.

# 4.3.5 Relocation of mobile facilities

One advantage of using mobile facilities suggested in the literature is relocation of facilities. The levelised delivered cost, when using both types of mobile facility, is lowest when the facilities are relocated approximately 210 times (i.e. marginally under once per month for a 20 year operation). Relocating more frequently than this results in commodity transport cost savings being offset by the cost of continuously relocating each mobile facility. The average woodchip transport distance to each mobile facility is 9.60 km, 2.76 km and 1.96 km when mobile facilities are relocated 19 (annually), 240 (monthly) and 480 (every two weeks) times over the 20 year operation, respectively. The associated levelised delivered cost for a mobile torrefaction facility is 8.58  $GJ^{-1}$ , 8.45  $GJ^{-1}$  and 8.48  $GJ^{-1}$  (Figure 4-5).

# 4.3.6 Point-of-delivery

Figure 4-6 shows how the levelised delivered cost varies when the bio-fuel facility is located at an additional transport distance from the forest harvest region (scenario B) for an annual harvest of 1.717 million m<sup>3</sup> (equivalent to delivering 2000 ODTPD woodchips). Woodchip delivery is the lowest cost pathway for additional distances under approximately 50 km (base value haul costs). For greater distances, torrefied wood is the lowest cost pathway for delivering the forest residue resource. Bio-slurry is a less costly delivery pathway than woodchips at distances above 360 km, although the delivered cost of torrefied wood is still lower. Further analysis shows that if



**Figure 4-5** Variation in levelised delivered cost of torrefied wood when relocating mobile torrefaction facilities.

woodchip haul costs are as low as 0.85 \$ km<sup>-1</sup>, delivering woodchips is the lowest cost pathway for additional distances up to at least 500 km.

Figure 4-7 shows the lowest cost delivery pathway for a range of annual harvests and additional transport distances. Bio-oil is always the most expensive pathway for delivering the forest residue resource (for the analysis range considered) followed by bio-slurry. Torrefied wood is a less costly pathway than both bio-oil and bio-slurry, and is also less costly than woodchips depending on the annual harvest volume and additional transport distance required. In Figure 4-7, implementing mobile torrefaction facilities is shown to be the lowest cost delivery pathway (for any annual harvest volume) when the point of delivery requires additional transport distances of more than 250 km. When there is no additional transport requirement, torrefied wood is the lowest cost delivery pathway when annual harvests are above 2.25 million m<sup>3</sup>.



**Figure 4-6** Levelised delivered cost when additional transport distances to a bio-fuel facility are required (scenario B; annual harvest of 1.717 million m<sup>3</sup>).



**Figure 4-7** Visualisation of lowest cost delivery option for a range of harvest volumes and additional transport distances.

# 4.4 **Bio-fuel production**

This Section presents results for each bio-fuel production route using feedstock from each feedstock delivery pathway, where applicable (Figure 3-4). Petrol and diesel fuel production (Section 4.4.1) is analysed separately from hydrogen production (Section 4.4.2). Analysis includes levelised costs of bio-fuel production for a range of forest residue harvest volumes, which provides insight into optimal harvest volumes for each bio-fuel production route. The effect of additional transport distances is also examined for when the bio-fuel facility is located outside of the forested region (scenario B), and finally the effect of bio-fuel production parameters on bio-fuel costs are discussed further, given their influence on LCB results as highlighted in the sensitivity study.

Note: Results shown throughout this Section relate to bio-fuel facilities located at the centre of the forested region (scenario A), except when discussing point-of-delivery distances in Section 4.4.1.2 and Section 4.4.2.2.

#### 4.4.1 Petrol and diesel

## 4.4.1.1 Optimal harvest volumes

Figure 4-8 shows how the levelised cost of bio-fuel (LCB) for petrol and diesel fuels varies with the annual harvest volume of forest residues, for both Fischer-Tropsch (FT) and upgrading bio-fuel production routes.

LCB results all follow the same trend: for small annual harvest volumes the LCB is high because there are large expenditures for the bio-fuel facility, yet the quantity of fuel produced is small; for large annual harvest volumes the LCB is high because, although more fuel is produced and economies of scale reduce the cost of the bio-fuel facility (per tonne feed), transport costs are high due to harvesting forest residues over an increasingly large area. There is an optimum low cost of bio-fuel production between these two regions.

The minimum cost of FT bio-fuels produced from a woodchip feedstock occurs at an annual harvest of 992,000 m<sup>3</sup> forest residues (equivalent to a 1160 ODTPD bio-fuel



Figure 4-8 Levelised cost of bio-fuel (petrol and diesel) for a range of annual harvest volumes.



**Figure 4-9** Levelised cost of bio-fuel (petrol and diesel) for optimally sized bio-fuel facilities when additional transport distances to a bio-fuel facility are required (scenario B).

facility), and the LCB is 1.39 \$ litre<sup>-1</sup>. Transport requirements are reduced for delivery pathways using mobile facilities, therefore minimum LCB results for bio-fuel facilities using bio-oil, bio-slurry or torrefied wood occur at larger annual harvest volumes (i.e. larger bio-fuel facilities). Optimum sizes of FT facilities using either bio-oil, bio-slurry or torrefied wood feedstock occur for harvests of 4.30 million m<sup>3</sup> (2750 TPD), 3.71 million m<sup>3</sup> (2700 TPD), and 2.84 million m<sup>3</sup> (2250 TPD), respectively. For bio-fuel production via gasification and FT synthesis, woodchip feedstock provides the lowest LCB. However, if woodchip haul costs increase, torrefied wood feedstock becomes the lowest cost option, for larger annual harvests. For example, if haul costs of woodchips increases to 3 \$ km<sup>-1</sup>, torrefied wood feedstock produces the lowest LCB results for annual harvests above approximately 1 million m<sup>3</sup>, as shown in Figure 4-8.

Bio-fuel production costs via bio-oil upgrading are lower than those produced via FT synthesis, and are less influenced by annual harvest volume as mobile fast pyrolysis facilities reduce transport requirements of bio-oil feedstock delivery. The minimum LCB occurs at 4.30 million m<sup>3</sup> (2750 TPD), 3.13 million m<sup>3</sup> (2000 TPD) and 1.96 million m<sup>3</sup> (1250 TPD) for bio-oil upgrading facilities using hydrogen sourced from SMR, bio-oil steam reformation and purchased hydrogen (1.5 kg<sup>-1</sup>), respectively. The minimum LCB for producing bio-fuels via upgrading is 0.86 litre<sup>-1</sup> when using SMR for the hydrogen source.

The cost of purchasing or producing hydrogen for upgrading bio-oil affects final bio-fuel production costs. The effect of purchased hydrogen prices  $(1.5 - 6 \ kg^{-1})$  are shown in Table 4-5. If SMR is used to produce hydrogen from methane, the price of methane has little effect on LCB - a 50% change in methane price results in a 3.6% change in LCB, according to sensitivity study results. However, the capital and operating costs of a SMR facility play a larger role in dictating LCB results. Similarly, when hydrogen is sourced from a bio-oil steam reformation facility, the LCB of upgrading is influenced by the capital and operating costs of the steam reformation facility. However, the largest impact on LCB results for upgrading occurs when bio-fuel production parameters are varied (i.e. the hydrogen requirement for upgrading bio-fuels, or hydrogen yields from steam reformation of bio-oil). These are discussed further in Section 4.4.1.3.

Hydrogen purchase price (\$ kg <sup>-1</sup> )	LCB at optimum harvest (\$ litre <sup>-1</sup> )		
1.5	0.88		
3	1.05		
4.5	1.22		
6	1.39		

 Table 4-5 Levelised costs of bio-fuels via upgrading bio-oil using hydrogen purchased from an external source.

## 4.4.1.2 Point-of-delivery

Figure 4-9 shows how LCB results for petrol and diesel vary as the additional distance to the bio-fuel facility from the forested region is increased from 0 to 500 km (scenario B). To compare minimum bio-fuel production costs for each bio-fuel production route, LCB results shown in Figure 4-9 are for optimally sized bio-fuel facilities (Section 4.4.1.1). The optimal size of bio-fuel facility for each bio-fuel production route does not change with additional transport distance because additional transport distances are independent of the annual harvest volume of forest residues.

For a Fischer-Tropsch bio-fuel facility, woodchip feedstock produces the lowest cost bio-fuels for distances up to 290 km. For distances greater than 250 km, torrefied wood feedstock produces the lowest cost bio-fuels for a FT bio-fuel facility. Upgrading bio-oil provides the lowest LCB result of all petrol and diesel production routes for all additional transport distances. The final cost of bio-fuel does not increase substantially with distance because bio-oil transportation is inexpensive (compared to woodchip or torrefied wood commodities).

#### 4.4.1.3 Bio-fuel production parameters

Parameters that influence bio-fuel production via Fischer-Tropsch synthesis are the quantity of bio-fuels produced per tonne feed, and also the fraction of bio-fuel that is petrol or diesel. There are numerous variables throughout gasification and FT synthesis (e.g. temperature, pressure, use of catalysts) that can influence the quantity of bio-fuel produced, which are beyond the scope of this study. These variables have been grouped into one parameter for bio-fuel production (Section 3.3.2), and has been applied to all feedstock types. The effect of this parameter is not complex - if a smaller amount of bio-fuel is produced per tonne feedstock the LCB increases (Equation 3.66), and vice versa. The fraction of either petrol or diesel produced by Fischer-Tropsch synthesis does impact LCB results as significantly. For example, a FT bio-fuel production facility optimally sized for woodchip feedstock will produce bio-fuels between a LCB of 1.27 \$ litre<sup>-1</sup> to 1.48 \$ litre<sup>-1</sup> as the fraction of fuel ranges from 100% petrol to 100% diesel, respectively. Producing 100% petrol or diesel is not practical at FT facilities, although this result demonstrates the impact of this variable in the model.

Parameters that influence bio-fuel production via bio-oil upgrading are the quantity of bio-fuels produced per tonne feed, the fraction of bio-fuel that is petrol or diesel and, when a portion of the bio-oil feed is steam reformed to produce the hydrogen required for upgrading, the quantity of hydrogen required per tonne bio-oil feed and the hydrogen yield of bio-oil steam reformation.

The parameters that influence the total quantity and fractions of petrol or diesel products from upgrading are set up in the model with the same architecture as those for FT synthesis. However, the reference for upgraded fuels produces 95% petrol and 5% diesel (as opposed to 40% petrol and 60% diesel in Fischer-Tropsch synthesis). Furthermore, bio-fuel yields via bio-oil upgrading are subject to variation as this study has assumed that bio-oil methanol mixtures yield similar quantities of bio-fuels to when pure bio-oil is upgraded.

The hydrogen requirement of upgrading only influences the quantity of bio-fuels produced when upgrading uses hydrogen sourced from steam reformation of the bio-oil feed (when hydrogen is produced from SMR or purchased, the quantity of bio-fuel product remains the same although the size and cost of the SMR facility, or hydrogen purchase expenses, may change). For upgrading using hydrogen sourced from steam reformation of bio-oil, LCB results can vary due to steam reformation yields as well as upgrading hydrogen requirements (Figure 4-10). Results are more sensitive to the hydrogen yield of bio-oil steam reformation, and the LCB can increase from 0.90 \$ litre<sup>-1</sup> to 1.20 \$ litre<sup>-1</sup>.



**Figure 4-10** Levelised cost of bio-fuel (petrol and diesel) for ranges of hydrogen requirements for upgrading and hydrogen production from bio-oil steam reformation provided in the literature.

# 4.4.2 Hydrogen

#### 4.4.2.1 Optimal harvest volumes

Figure 4-11 shows how the levelised cost of bio-fuel (LCB) for hydrogen varies with the annual harvest volume of forest residues, for bio-fuel production routes of gasification followed by water gas shift reactions and bio-oil steam reformation. The minimum LCB of hydrogen produced from gasification and water gas shift (WGS) of a woodchip feedstock occurs for an annual harvest of 947,000 m<sup>3</sup> forest residues (equivalent to a 1100 ODTPD bio-fuel facility), and the LCB is 2.92 \$ kg<sup>-1</sup>. Optimum sizes of gasification and WGS facilities using either bio-oil, bio-slurry or torrefied wood feedstock occur for harvests of 4.01 million m<sup>3</sup> (2560 TPD) , 3.42 million m<sup>3</sup> (2490 TPD), 2.84 million m<sup>3</sup> (2250 TPD), respectively. For hydrogen production via gasification and WGS, woodchip feedstock provides the lowest LCB. However, as haul



Figure 4-11 Levelised cost of bio-fuel (hydrogen) for a range of annual harvest volumes.



**Figure 4-12** Levelised cost of bio-fuel (hydrogen) for optimally sized bio-fuel facilities when additional transport distances to a bio-fuel facility are required (scenario B).

costs increase, torrefied wood feedstock becomes the lowest cost option, for larger annual harvests.

Hydrogen production costs via steam reformation of bio-oil are competitive with those of gasification and WGS of a woodchip feedstock. LCB results for steam reformation of bio-oil are less influenced by annual harvest volume, as mobile fast pyrolysis facilities reduce transport requirements of feedstock delivery. The minimum LCB occurs at an annual harvest of 2.84 million  $m^3$  (1810 TPD), and the cost of hydrogen production is 3.25 \$ kg<sup>-1</sup>.

# 4.4.2.2 Point-of-delivery

Figure 4-12 shows how LCB results for hydrogen production vary as the additional distance to the bio-fuel facility from the forested region is increased from 0 to 500 km (scenario B). To compare minimum bio-fuel production costs for each bio-fuel production route, LCB results shown in Figure 4-12 are for optimally sized bio-fuel facilities (Section 4.4.2.1).

For a gasification and WGS bio-fuel facility, woodchip feedstock produces the lowest cost hydrogen for distances up to 295 km. For larger distances, torrefied wood feedstock produces the lowest cost hydrogen. Steam reformation of bio-oil produces lowest LCB results of all hydrogen bio-fuel production routes for all additional transport distances above 85 km.

# 4.4.2.3 Bio-fuel production parameters

Hydrogen production yield via gasification and WGS varies depending on the set up of the gasification reactor and also the conditions of the water gas shift reactions. For an optimally sized gasification and WGS facility utilising woodchip feedstock the levelised cost of hydrogen varies between 2.74 \$ kg<sup>-1</sup> and 3.22 \$ kg<sup>-1</sup> as the production yield parameter is varied between maximum and minimum values available in the literature.

Hydrogen production via steam reformation is still in development, although implementing production rates from the literature produces LCB results between  $3.14 \text{ kg}^{-1}$  and  $3.25 \text{ kg}^{-1}$ , for an optimally sized bio-oil steam reformation facility.

# 4.5 Other considerations

#### 4.5.1 Forest residue resource

The model presented in this study provides an analysis of using mobile facilities to harvest a general forest residue resource. However, biomass is a spatially and temporally dependent resource, and further considerations are necessary when examining particular geographic regions. Transport distances, hauling costs and sites available for mobile facilities require knowledge of the availability and location of the forest residue resource, as well as the local road network. These considerations would also prove useful when investigating the impact and cost savings possible from the relocation of mobile facilities.

## 4.5.2 Market demand and marginal costs

The market demand, or competition, for forest residue resources may impact the quantity of resource available for bio-fuel production. As the demand for forest residues increases the purchase price of forest residues will likely increase, and industries that can harvest the resource for lowest costs will be in the best position for securing the resource. In this respect, competition for forest residues is generally not a concern of physical availability but instead an economic matter, in which demand change from one user can cause significant price changes for other users [27].

The marginal cost of using forest residue resources may reduce the extent to which the resource is utilised. The marginal cost of harvesting residues increases at a faster pace than the marginal cost of roundwood [28]. Therefore, harvesting large volumes of forest residues for large size bio-fuel facilities may not occur. Rather, supply may switch to roundwood when feedstock costs of roundwood are cheaper than that of forest residues. At this point, however, the bio-fuel sector starts to infringe on the supply of roundwood used by the logging industry and there is increased competition between industries.

On the other hand, mobile facilities reduce the marginal cost of a forest residue resource as they reduce transport requirements, which is an important factor of marginal costs [27]. Therefore, mobile facilities may allow greater penetration of forest residue

resources that are largely unused due to high marginal costs [28]. Integrating forest residue recovery (and mobile facility use) with existing logging industry could alleviate negative consequences of competition between bio-fuel and logging sectors. The costs of harvesting forest residues are expected to decrease as labour and equipment could be shared.

# 4.5.3 Mobile facility configurations

The mobile facility configurations used in this study for fast pyrolysis and torrefaction represent those in the literature. This study has not examined the influence of varying mobile facility configurations (i.e. varying product yields produced by each process). This would likely require experimental data and/or detailed process simulation, as the quantity and characteristics of the products, as well as the thermal and electrical requirements of the mobile facility are expected to change for each different set up.

Configuration of mobile facilities for optimal product yields would be beneficial for decreasing the levelised delivered cost of mobile facility products. Ensuring sufficient syngas is produced to meet drying and thermal requirements of the facilities would remove the costs of purchasing propane. However, increasing syngas production would likely decrease the yield of other products to be delivered to the bio-fuel facility, thus increasing the levelised delivered cost. Essentially, the optimum set up of mobile facilities would ensure that all thermal and electrical demands are met, whilst also producing maximum yields of products that will be used as feedstock at bio-fuel facilities.

# 4.5.4 Sustainability of bio-fuel production from forest residues

Utilisation of any biomass resource requires a sustainable approach that encompasses environmental, social and economic issues [80]. One fundamental concept of environmental sustainability is to ensure a lasting supply of resource (i.e. harvest rates should not exceed the regeneration rates) [81]. In terms of a forest residue resource this means that, in conjunction with other logging activities, the rate at which forests are harvested should not exceed forest regeneration rates. This issue is generally managed by policies such as annual allowable cuts, which limit the amount of logging that can occur within a specific region each year. However, removal of forest residues from logging sites raises concerns related to soil productivity and biodiversity, that could affect the growth rate of new forest [82]. If forest residues are removed as woodchips, then the majority of nutrients within the wood will be removed as part of the woodchip feedstock. When mobile facilities are implemented, it has been proposed that solid products of mobile facilities (e.g. bio-char) could be re-distributed back into soils at forestry sites as an option for soil remediation after removal of forest residues (e.g. [83]). However, any redistribution of products to the logging fields would incur additional costs.

Social aspects of sustainability have not been addressed in this study, as they are very dependent on the region or country in which such bio-fuel production systems would be implemented.

The cost analysis presented in this study provides a starting point to discuss economic issues related to the use of mobile facilities and utilisation of forest residue resources, yet insight into sustainable economic growth would require details of regional and national markets of both resource extraction and alternative fuel options (e.g. fossil fuels), which are also temporally and location dependent.

In general, continuous examination of bio-energy systems should be performed to ensure that no adverse environmental effects arise from forest residue removal at logging sites, and social and economic factors should be investigated for regions where forest residue resources are to be utilised.

#### 4.5.5 Initiating use of mobile facilities

Since mobile facilities are not commercially available and are still subjects of research, it is unlikely that the first use of mobile facilities will be on a large scale, as the risk of purchasing multiple mobile facilities at one time to meet an entire harvest requirement of a bio-fuel facility is high. However, results show that small numbers of mobile facilities do not generally create lower delivered feedstock costs compared to conventional woodchip delivery, and, in addition, bio-fuel facilities using feedstock from mobile facility pathways produce lowest cost fuels for large annual harvests (which would require many mobile facilities to meet harvest requirements). Thus, one question is whether a gradual introduction of mobile facilities can be economically viable. This

study does not provide results that can provide a complete answer to this question, although the feasibility would likely depend on the end use of the forest residue resource and the availability of existing infrastructure that could make use of mobile facility commodities as feedstock (which may not be bio-fuel facilities).

For small scale implementation to be economically feasible, transport distances need to be large (over 250 km according to the results of this study). Therefore, mobile facilities would be most favourable for harnessing forest residue resources that are too far from existing facilities that would use such resources to justify conventional woodchip delivery. Mobile facilities would initially be suited to existing infrastructure that could make use of the commodities they produce without needing major changes to current operations. For instance, mobile facilities could produce either bio-slurry or torrefied wood to be co-fired with coal at existing power stations [13], [39]. Over time, additional mobile facilities could be purchased to access greater volumes of forest residues and the destination of mobile facility commodities may switch to a bio-fuel facility purpose-built to access feedstock from mobile facility delivery pathways.

# 4.6 Summary

This Chapter has provided results and discussion of the use of mobile facilities in harvesting a forest residue resource for bio-fuel production.

Of the four pathways considered to deliver a forest residue resource to a bio-fuel facility, woodchip delivery is the lowest cost option when annual harvests are small or transport distances are short. Implementing mobile facilities becomes economically attractive for larger harvests or when additional transport distances are required.

Reductions in levelised delivered cost for mobile facility pathways depend on (i) the purchase and operating costs of the mobile facilities (ii) the energy content of the products of mobile facilities (i.e. the conversion efficiencies of the mobile facilities), and (iii) the cost reductions of transportation when compared to conventional woodchip transport. Torrefied wood is the lowest cost pathway of delivering a forest residue resource when using mobile facilities, and is competitive with woodchip delivery at larger harvests and transport distances. Levelised delivered costs are particularly sensitive to densities of commodities delivered to bio-fuel facilities, transport factors, and, when mobile facilities are implemented, mobile facility capital costs, thermal and electrical demands of mobile facilities, and the initial moisture content of forest residues.

Results for levelised cost of bio-fuels show that there are optimal harvest volumes (bio-fuel facility sizes) for each bio-fuel production route, which yield minimum bio-fuel production costs. These occur primarily as the benefit of economies of scale for larger bio-fuel facilities competes against increasing transport costs for large harvest volumes. Optimal harvest volumes are larger for bio-fuel production routes that use feedstock sourced from mobile facilities, as mobile facilities reduce total transport requirements.

Petrol and diesel production costs are lowest for bio-fuel production routes that upgrade bio-oil feedstock, even though the levelised delivered cost of bio-oil is highest compared against other delivery pathways. This is mainly because greater quantities of bio-fuels are produced from the upgrading process, compared to Fischer-Tropsch synthesis. The levelised cost of bio-fuels produced via Fischer-Tropsch synthesis are lowest for woodchip feedstock, unless haul costs or transport distances are large, in which case torrefied wood feedstock provides the lowest cost of bio-fuel.

Hydrogen production via steam reformation of bio-oil is competitive with gasification and water gas shift processes for hydrogen production, particularly for larger annual harvest of forest residues. Hydrogen production costs from gasification and WGS processes are lowest for woodchip feedstock, unless transport costs are high, when torrefied wood feedstock becomes the lowest cost option. Levelised costs of bio-fuels are particularly sensitive to bio-fuel facility costs and bio-fuel production parameters.

If a bio-fuel facility is located outside of the forested region, bio-fuel production processes that use mobile facility commodities as feedstock provide lowest cost bio-fuels as the additional transport distance is increased.

This study has provided an analysis of the use of mobile facilities to harvest a forest residue resource for bio-fuel production. Mobile facility implementation for bio-fuel production in a specific region or country will require detailed information of the local forest residue resource, local resource and fuel markets, and local infrastructure.

Extensive research into environmental, social and economic issues relevant to the specific region is also necessary to ensure a sustainable supply of bio-fuels.

The following Chapter presents concluding remarks from this study as well as suggestions for further work.

# 5 Conclusions

The objective of this study was to investigate the technical and economic implications of producing bio-fuels when using a network of mobile facilities to deliver a forest residue resource to a bio-fuel facility. To meet this objective, this study has developed a general model for examining the use of mobile facilities to deliver a forest residue resource for bio-fuel production.

Analyses investigated the levelised delivered cost of a forest residue resource using four pathways to deliver the resource as either woodchip, bio-oil, bio-slurry or torrefied wood. The levelised cost of bio-fuels was calculated for bio-fuel processes using each type of feedstock. Bio-fuel production processes analysed in this study were gasification followed by either Fischer-Tropsch synthesis or water gas shift reactions, and upgrading or steam reformation of bio-oil.

Results show that implementing a network of mobile facilities will reduce the energy content of forest residues delivered to a bio-fuel facility as mobile facilities use a fraction of the biomass energy content to meet thermal or electrical demands. The total energy delivered by bio-oil, bio-slurry and torrefied wood is 45%, 65% and 87% of the initial forest residue energy content, respectively. Nonetheless, implementing mobile facilities is economically feasible when large transport distances are required. Transport costs are reduced to less than 40% of total levelised delivered costs when mobile facilities are implemented, compared against 80% of costs for conventional woodchip delivery, for annual harvests of 1.717 million m<sup>3</sup> (equivalent to 2000 ODTPD). Torrefied wood is the lowest cost pathway of delivering a forest residue resource when using mobile facilities. Cost savings occur against woodchip delivery for annual forest residue harvests above 2.25 million m<sup>3</sup> or when transport distances greater than 250 km are required.

The minimum cost of petrol and diesel production is 0.86 \$ litre<sup>-1</sup> when bio-oil is upgraded using hydrogen sourced from SMR. The optimum size of this facility is 2750 TPD and requires an annual harvest of 4.30 million m<sup>3</sup>. The minimum cost of hydrogen production is 2.92 \$ kg<sup>-1</sup> via the gasification of a woodchip feedstock and

subsequent water gas shift reactions. The optimum size for this facility is 1100 ODTPD and requires an annual harvest of 947,000 m<sup>3</sup>.

The sensitivity study performed has shown important parameters that significantly impact results:

- initial moisture content of forest residues
- mobile facility capital cost
- mobile facility electrical and drying requirements
- densities of commodities (woodchip, bio-oil, bio-slurry and torrefied wood)
- transport factors (truck volumes, haul costs, and tortuosity factors)
- bio-fuel facility capital cost
- bio-fuel production factors (bio-fuel yields per unit feedstock)

This study has provided a general analysis of mobile facility use in delivering a forest residue resource for bio-fuel production. Mobile facility implementation for bio-fuel production in a specific region or country will require detailed information of the local forest residue resource, local resource and fuel markets and also extensive research into environmental, social and economic issues relevant to the specific region to ensure a sustainable supply of bio-fuels. To this end, recommendations of further work extending this research are presented below.

# 5.1 Recommendations

The following recommendations offer both improvements to the model presented in this study and topics of further research, which could enhance the findings of this study and/or provide useful techniques to assist in the application of the model to specific regions of interest:

- use experimental data or detailed process modelling to examine mobile facility product yields and characteristics for different mobile facility configurations
- update financial data for mobile torrefaction facilities if and when it becomes available in the literature

- investigate the extent to which mobile facilities could be integrated with current logging practices
- extend the model framework to include the use of geographical information systems (GIS) software (e.g. [66]), which could:
  - allow for definition of a forest residue resource that includes spatial and temporal variations
  - provide detailed road network information to enhance calculations of transport distances and costs
  - determine suitable sites for mobile facility locations, which in turn will provide more accurate results for cost savings that may occur through relocation of facilities
  - determine optimal sites and sizes for bio-fuel facilities
  - include social and economic factors when considering utilisation of forest residue resources and also locations of mobile and bio-fuel facilities

# 6 References

- A. J. Liska and R. K. Perrin, "Indirect land use emissions in the life cycle of biofuels : regulations vs science," *Biofuels, Bioproducts & Biorefining*, no. 3, pp. 318–328, 2009.
- [2] A. Kendall, B. Chang, and B. Sharpe, "Accounting for time-dependent effects in biofuel life cycle greenhouse gas emissions calculations," *Environmental Science and Technology*, vol. 43, no. 18, pp. 7142–7, Sep. 2009.
- [3] J. M. DeCicco, "Biofuels and carbon management," *Climate Change*, vol. 111, no. 3–4, pp. 627–640, Jul. 2011.
- [4] W. Hallgren, C. A. Schlosser, E. Monier, D. Kicklighter, A. Sokolov, and J. Melillo, "Climate impacts of a large-scale biofuels expansion," *Geophysical Research Letters*, vol. 40, no. 8, pp. 1624–1630, Apr. 2013.
- [5] G. R. Timilsina and S. Mevel, "Biofuels and Climate Change Mitigation: A CGE Analysis Incorporating Land-Use Change," *Environmental and Resource Economics*, vol. 55, no. 1, pp. 1–19, Oct. 2012.
- [6] B. Lippke, R. Gustafson, R. Venditti, P. Steele, T. A. Volk, E. Oneil, L. Johnson, M. E. Puettmann, and K. Skog, "Comparing Life-Cycle Carbon and Energy Impacts for Biofuel, Wood Product, and Forest Management Alternatives," *Forests Products Journal*, vol. 62, no. 4, pp. 247–257, 2012.
- [7] A. Sultana and A. Kumar, "Optimal configuration and combination of multiple lignocellulosic biomass feedstocks delivery to a biorefinery," *Bioresource Technology*, vol. 102, no. 21, pp. 9947–9956, Nov. 2011.
- [8] P. C. Badger and P. Fransham, "Use of mobile fast pyrolysis plants to densify biomass and reduce biomass handling costs - A preliminary assessment," *Biomass* and Bioenergy, vol. 30, no. 4, pp. 321–325, Apr. 2006.
- [9] Renewable Fuel Technologies, "Mobile Torrefaction A Scalable Biomass to Energy Solution," 2011. [Online]. Available: http://renewablefueltech.files.wordpress.com/2011/11/es-rft-9-16-2011-comp.pdf. [Accessed: 15-Jul-2013]
- [10] C. B. Sorenson, "A Comparitive Financial Analysis of Fast Pyrolysis Plants in Southwest Oregon," University of Montana, 2010.

- [11] A. V. Bridgwater, "Review of fast pyrolysis of biomass and product upgrading," *Biomass and Bioenergy*, vol. 38, pp. 68–94, Mar. 2012.
- B. Acharya, I. Sule, and A. Dutta, "A review on advances of torrefaction technologies for biomass processing," *Biomass Conversion and Biorefinery*, vol. 2, no. 4, pp. 349–369, Sep. 2012.
- P. C. A. Bergman, A. R. Boersma, R. W. R. Zwart, and J. H. A. Kiel,
   "Torrefaction for biomass co-firing in existing coal-fired power stations," Utrecht, The Netherlands: Energy Research Centre of the Netherlands (ECN), 2005.
- [14] D. Paré, P. Bernier, E. Thiffault, and B. D. Titus, "The potential of forest biomass as an energy supply for Canada," *The Forestry Chronicle*, vol. 87, no. 1, pp. 71– 76, 2011.
- [15] P. Pelkonen, P. Hakkila, T. Karjalainen, and B. Schlamadinger, Eds., Woody Biomass as an Energy Source – Challenges in Europe, Joensuu, Finland: European Forest Institute, 2001.
- [16] S. B. Jones, C. Valkenburg, C. W. Walton, D. C. Elliott, J. E. Holladay, D. J. Stevens, C. Kinchin, and S. Czernik, "Production of Gasoline and Diesel from Biomass via Fast Pyrolysis, Hydrotreating and Hydrocracking: A Design Case," Richland, Washington, United States: Pacific Northwest National Laboratory (PNNL), 2009.
- [17] T. Vispute, "Pyrolysis Oils : Characterization, Stability Analysis, and Catalytic Upgrading to Fuels and Chemicals," University of Massachusetts, 2011.
- [18] D. Rennard, R. French, S. Czernik, T. Josephson, and L. Schmidt, "Production of synthesis gas by partial oxidation and steam reforming of biomass pyrolysis oils," *International Journal of Hydrogen Energy*, vol. 35, no. 9, pp. 4048–4059, May 2010.
- [19] S. Sarkar and A. Kumar, "Large-scale biohydrogen production from bio-oil," *Bioresource Technology*, vol. 101, no. 19, pp. 7350–61, Oct. 2010.
- [20] Industrial Forestry Service Ltd, M.D.T. Ltd, and Murray Hall Consulting Ltd, "Wood Based Biomass Energy Potential of British Columbia: Prepared for BC Hydro's Integrated Resource Planning Process," 2010.
- [21] H. Harrill, "Costs and Productivity of Woody Biomass Harvesting in Integrated Stand Conversion and Residue Recovery Operations," Humbolt State University, 2010.
- [22] BIOCAP Canada, "An Information Guide on Pursuing Biomass Energy Opportunities and Technologies in British Columbia," 2008.

- [23] D. Ciolkosz, "Characteristics of Biomass as a Heating Fuel," 2010. [Online]. Available: http://pubs.cas.psu.edu/freepubs/pdfs/ub043.pdf. [Accessed: 25-Jul-2013].
- [24] J. G. Speight, "Properties of Fuels from Wood Sources," in *The Biofuels Handbook*, Royal Society of Chemistry, 2011, pp. 304 – 330.
- [25] P. Ralevic, M. Ryans, and D. Cormier, "Assessing forest biomass for bioenergy: Operational challenges and cost considerations," *The Forestry Chronical* vol. 86, no. 1, pp. 43–50, 2010.
- [26] S. T. Grushecky, J. Wang, and D. W. McGill, "Influence of site characteristics and costs of extraction and trucking on logging residue utilization in southern West Virginia," *Forest Products Journal*, vol. 57, no. 7, pp. 63–67, Jul. 2007.
- [27] E. Bjørnstad, "An engineering economics approach to the estimation of forest fuel supply in North-Trøndelag county, Norway," *Journal of Forest Economics*, vol. 10, no. 4, pp. 161–188, Feb. 2005.
- [28] R. Lundmark, "Cost structure of and competition for forest-based biomass," *Scandinavian Journal of Forest Research*, vol. 21, no. 3, pp. 272–280, Jun. 2006.
- [29] D. Mohan, C. U. Pittman, and P. H. Steele, "Pyrolysis of Wood/Biomass for Biooil: A Critical Review," *Energy & Fuels*, vol. 20, no. 3, pp. 848–889, 2006.
- [30] J. G. Rogers and J. G. Brammer, "Estimation of the production cost of fast pyrolysis bio-oil," *Biomass and Bioenergy*, vol. 36, pp. 208–217, Jan. 2012.
- [31] S. Xiu and A. Shahbazi, "Bio-oil production and upgrading research: A review," *Renewable and Sustainable Energy Reviews*, vol. 16, no. 7, pp. 4406–4414, Sep. 2012.
- [32] E. Butler, G. Devlin, D. Meier, and K. McDonnell, "A review of recent laboratory research and commercial developments in fast pyrolysis and upgrading," *Renewable and Sustainable Energy Reviews*, vol. 15, no. 8, pp. 4171–4186, Oct. 2011.
- [33] S. Thangalazhy-Gopakumar, S. Adhikari, H. Ravindran, R. B. Gupta, O. Fasina, M. Tu, and S. D. Fernando, "Physiochemical properties of bio-oil produced at various temperatures from pine wood using an auger reactor," *Bioresource Technology*, vol. 101, no. 21, pp. 8389–95, Dec. 2010.
- [34] F. Preto, "Current Issues and Future Directions in Pyrolysis of Biomass," 2005.
   [Online]. Available: http://www.biocap.ca/images/pdfs/conferenceSpeakers/Preto\_F.pdf.
   [Accessed: 15-Jul-2013].

- [35] P. C. Badger, S. Badger, M. E. Puettmann, P. Steele, and J. Cooper, "Techno-Economic Analysis: Preliminary Assessment of Pyrolysis Oil Production Costs and Material Energy Balance Associated with a Transportable Fast Pyrolysis System," *BioResources*, vol. 6, no. 1, pp. 34–47, 2011.
- [36] A. Oasmaa and C. Peacocke, "A guide to physical property characterisation of biomass- derived fast pyrolysis liquids," Espoo, Finland: Technical Research Centre of Finland (VTT), 2001.
- [37] M. L. Honkela, T.-R. Viljava, A. Gutierrez, A. Outi, and I. Krause, "Hydrotreating for Bio-Oil Upgrading," in *Thermochemical Conversion of Biomass to Liquid Fuels and Chemicals*, M. Crocker, Ed. Royal Society of Chemistry, 2010, pp. 288– 306.
- [38] K. Raffelt, E. Henrich, A. Koegel, R. Stahl, J. Steinhardt, and F. Weirich, "The BTL2 Process of Biomass Utilization Entrained-Flow Gasification of Pyrolyzed Biomass Slurries," *Applied Biochemistry and Biotechnology*, vol. 129–132, pp. 153–164, 2006.
- [39] H. Wu, Y. Yu, and K. Yip, "Bioslurry as a Fuel. 1. Viability of a Bioslurry-Based Bioenergy Supply Chain for Mallee Biomass in Western Australia," *Energy & Fuels*, vol. 24, no. 10, pp. 5652–5659, Oct. 2010.
- [40] N. Dahmen, E. Henrich, E. Dinjus, and F. Weirich, "The bioliq® bioslurry gasification process for the production of biosynfuels, organic chemicals, and energy," *Energy, Sustainability and Society*, vol. 2, no. 3, 2012.
- [41] J. J. Chew and V. Doshi, "Recent advances in biomass pretreatment Torrefaction fundamentals and technology," *Renewable and Sustainable Energy Reviews*, vol. 15, no. 8, pp. 4212–4222, Oct. 2011.
- [42] Integro Earth Fuels, "Integro Earth Fuels," 2013. [Online]. Available: http://www.integrofuels.com/press.html. [Accessed: 14-May-2013].
- [43] A. Uslu, A. P. C. Faaij, and P. C. a. Bergman, "Pre-treatment technologies, and their effect on international bioenergy supply chain logistics. Techno-economic evaluation of torrefaction, fast pyrolysis and pelletisation," *Energy*, vol. 33, no. 8, pp. 1206–1223, Aug. 2008.
- [44] B. Arias, C. Pevida, J. Fermoso, M. G. Plaza, F. Rubiera, and J. J. Pis, "Influence of torrefaction on the grindability and reactivity of woody biomass," *Fuel Processing Technology*, vol. 89, no. 2, pp. 169–175, Feb. 2008.

- [45] M. Balat, M. Balat, E. Kırtay, and H. Balat, "Main routes for the thermoconversion of biomass into fuels and chemicals. Part 2: Gasification systems," *Energy Conversion and Management*, vol. 50, no. 12, pp. 3158–3168, Dec. 2009.
- [46] J. Ciferno and J. J. Marano, "Benchmarking Biomass Gasification Technologies for Fuels, Chemicals and Hydrogen Production," National Energy Technology Laboratory (NETL), 2002.
- [47] R. M. Swanson, J. A. Satrio, R. C. Brown, A. Platon, and D. D. Hsu, "Techno-Economic Analysis of Biofuels Production Based on Gasification," Golden, Colorado, United States: National Renewable Energy Laboratory (NREL), 2010.
- [48] NREL, "Hydrogen Production Cost Estimate Using Biomass Gasification Independent Review," Golden, Colorado, United States: National Renewable Energy Laboratory (NREL), 2011.
- [49] S. Sarkar, A. Kumar, and A. Sultana, "Biofuels and biochemicals production from forest biomass in Western Canada," *Energy*, vol. 36, no. 10, pp. 6251–6262, Oct. 2011.
- [50] P. Spath, A. Aden, T. Eggeman, M. Ringer, B. Wallace, and J. Jechura, "Biomass to Hydrogen Production Detailed Design and Economics Utilizing the Battelle Columbus Heated Gasifier," Golden, Colorado, United States: National Renewable Energy Laboratory (NREL), 2005.
- [51] D. Volkmann, "Update on technology and projects," in *Gasification Technologies Conference*, 2004.
- [52] M. Sakaguchi, a. P. Watkinson, and N. Ellis, "Steam Gasification of Bio-Oil and Bio-Oil/Char Slurry in a Fluidized Bed Reactor," *Energy & Fuels*, vol. 24, no. 9, pp. 5181–5189, Sep. 2010.
- [53] Q. Chen, J. Zhou, B. Liu, Q. Mei, and Z. Luo, "Influence of torrefaction pretreatment on biomass gasification technology," *Chinese Science Bulletin*, vol. 56, no. 14, pp. 1449–1456, May 2011.
- [54] M. F. Bleeker, S. R. A. Kersten, and H. J. Veringa, "Pure hydrogen from pyrolysis oil using the steam-iron process," *Catalasis Today* vol. 127, pp. 278–290, 2007.
- [55] T. Malkow, "Novel and innovative pyrolysis and gasification technologies for energy efficient and environmentally sound MSW disposal," *Waste Management*, vol. 24, pp. 53–79, Jan. 2004.

- [56] F. Trippe, M. Fröhling, F. Schultmann, R. Stahl, E. Henrich, and A. Dalai, "Comprehensive techno-economic assessment of dimethyl ether (DME) synthesis and Fischer–Tropsch synthesis as alternative process steps within biomass-toliquid production," *Fuel Processing Technology*, vol. 106, pp. 577–586, Feb. 2013.
- [57] "NETL: Gasifipedia." [Online]. Available: http://www.netl.doe.gov/technologies/coalpower/gasification/gasifipedia/5-support/5-11\_ftsynthesis.html. [Accessed: 15-Jul-2013].
- [58] A. Kumar and S. Sarkar, "Bio-hydrogen Production from Bio-oil," in *Biofuels: Alternative Feedstocks and Conversion Processes*, A. Pandey, C. Larroche, S. Ricke, C. G. Dussap, and E. Gnansounou, Eds. 2011, pp. 485–486.
- [59] X. Hu and G. Lu, "Bio-oil steam reforming, partial oxidation or oxidative steam reforming coupled with bio-oil dry reforming to eliminate CO<sub>2</sub> emission," *International Journal of Hydrogen Energy*, vol. 35, no. 13, pp. 7169–7176, Jul. 2010.
- [60] J. A. Medrano, M. Oliva, J. Ruiz, L. García, and J. Arauzo, "Hydrogen from aqueous fraction of biomass pyrolysis liquids by catalytic steam reforming in fluidized bed," *Energy*, vol. 36, no. 4, pp. 2215–2224, Apr. 2011.
- [61] S. Ayalur Chattanathan, S. Adhikari, and N. Abdoulmoumine, "A review on current status of hydrogen production from bio-oil," *Renewable and Sustainable Energy Reviews*, vol. 16, no. 5, pp. 2366–2372, Jun. 2012.
- [62] R. J. Evans, S. Czernik, E. Chornet, C. J. Feik, R. French, and S. D. Phillips, "Engineering Scale Up of Renewable Hydrogen Production by Catalytic Steam Reforming of Peanut Shells Pyrolysis Products," Golden, Colorado, United States: National Renewable Energy Laboratory (NREL), 2002.
- [63] M. Wright and R. C. Brown, "Establishing the optimal sizes of different kinds of biorefineries," *Biofuels, Bioproducts and Biorefining*, vol. 1, pp. 191–200, 2007.
- [64] Y. Yu, J. Bartle, C. Z. Li, and H. Wu, "Mallee Biomass as a Key Bioenergy Source in Western Australia: Importance of Biomass Supply Chain," *Energy & Fuels*, vol. 23, no. 6, pp. 3290–3299, Jun. 2009.
- [65] R. P. Overend, "The Average Haul Distance and Transportation Work Factors for Biomass Delivered to a Central Plant," *Biomass*, vol. 2, pp. 75–79, 1982.
- [66] A. Sultana and A. Kumar, "Optimal siting and size of bioenergy facilities using geographic information system," *Applied Energy*, vol. 94, pp. 192–201, Jun. 2012.

- [67] M. Helin, "Moisture in wood fuels and drying of wood chips," 2005. [Online]. Available: http://www.northernwoodheat.net/htm/news/Finland/Symposiumpres/Dryingofwo odchips.pdf. [Accessed: 15-Jul-2013].
- [68] A. J. Toft, "A Comparison of Integrated Biomass to Electricity Systems," Aston University, 1996.
- [69] Diesel Service & Supply Inc, "Diesel Service & Supply Inc: Approximate Fuel Consumption Chart," 2013. [Online]. Available: http://www.dieselserviceandsupply.com/temp/Fuel\_Consumption\_Chart.pdf. [Accessed: 05-Apr-2013].
- [70] P. C. Badger, "Processing Cost Analysis for Biomass Feedstocks," Oak Ridge, Tennessee, United States: Oak Ridge National Laboratory (ORNL), 2002.
- [71] M. M. Wright, D. E. Daugaard, J. A. Satrio, and R. C. Brown, "Techno-economic analysis of biomass fast pyrolysis to transportation fuels," *Fuel*, vol. 89, pp. S2– S10, Nov. 2010.
- [72] D. C. Elliott, "Historical Developments in Hydroprocessing Bio-oils," *Energy & Fuels*, vol. 21, no. 3, pp. 1792–1815, May 2007.
- [73] C. N. Hamelinck, R. a. a. Suurs, and A. P. C. Faaij, "International bioenergy transport costs and energy balance," *Biomass and Bioenergy*, vol. 29, no. 2, pp. 114–134, Aug. 2005.
- [74] M. Wright, Y. Roman-Leshkov, and W. H. Green, "Investigating the technoeconomic trade-offs of hydrogen source using a response surface model of drop-in biofuel production via bio-oil upgrading," *Biofuels, Bioproducts and Biorefining*, no. 6, pp. 503–520, 2012.
- [75] M. M. Wright and R. C. Brown, "Distributed processing of biomass to bio-oil for subsequent production of Fischer-Tropsch liquids," *Biofuels, Bioproducts and Biorefining*, no. 2, pp. 229–238, 2008.
- [76] K. S. Ng and J. Sadhukhan, "Techno-economic performance analysis of bio-oil based Fischer-Tropsch and CHP synthesis platform," *Biomass and Bioenergy*, vol. 35, no. 7, pp. 3218–3234, Jul. 2011.
- [77] A. V. Bridgwater, A. J. Toft, and J. G. Brammer, "A techno-economic comparison of power production by biomass fast pyrolysis with gasification and combustion," *Renewable and Sustainable Energy Reviews*, vol. 6, no. 3, pp. 181–246, Sep. 2002.
- [78] J. F. Schulman, "Heavy Truck Weight and Dimension Limits in Canada," 2003.

- [79] T. Pootakham and A. Kumar, "Bio-oil transport by pipeline: a techno-economic assessment," *Bioresource Technology*, vol. 101, no. 18, pp. 7148–54, Sep. 2010.
- [80] Food and Agriculture Organiszation of the United Nations (FAO), "Criteria and indicators for sustainable woodfuels," Rome, Italy, 2010.
- [81] R. Goodland and H. Daly, "Environmental Sustainability: Universal and Non-Negotiable," *Ecological Applications*, vol. 6, no. 4, pp. 1002–1017, 1996.
- [82] P. Lamers, E. Thiffault, D. Paré, and M. Junginger, "Feedstock specific environmental risk levels related to biomass extraction for energy from boreal and temperate forests," *Biomass and Bioenergy*, vol. 55, pp. 212–226, Aug. 2013.
- [83] D. A. Laird, P. Fleming, D. D. Davis, R. Horton, B. Wang, and D. L. Karlen, "Impact of biochar amendments on the quality of a typical Midwestern agricultural soil," *Geoderma*, vol. 158, no. 3–4, pp. 443–449, Sep. 2010.
- [84] A. Kumar, J. B. Cameron, and P. C. Flynn, "Biomass power cost and optimum plant size in western Canada," *Biomass and Bioenergy*, vol. 24, pp. 445–464, 2003.
- [85] D. Bradley, "Canada Report on Bioenergy 2010," Ottowa, Canada, 2010.
- [86] H. Mahmudi and P. C. Flynn, "Rail vs truck transport of biomass," *Applied Biochemistry and Biotechnology*, vol. 129–132, pp. 88–103, Jan. 2006.
- [87] A. Sultana and A. Kumar, "Optimal configuration and combination of multiple lignocellulosic biomass feedstocks delivery to a biorefinery: Supplementary Information Section," *Bioresource Technology*, vol. 102, no. 21, pp. 9947–56, 2011.
- [88] Logistic Solution Builders Inc, "Operating Costs of Trucks In Canada," 2005.
   [Online]. Available: http://publications.gc.ca/collections/collection\_2011/tc/T46-14-2005-eng.pdf.
   [Accessed: 05-Apr-2013].
- [89] "Liquids Densities." [Online]. Available: http://www.engineeringtoolbox.com/liquids-densities-d\_743.html. [Accessed: 01-May-2013].
- [90] Dynamotive, "Dynamotive CQuest BioChar," Richmond, Canada: Dynamotive Information Booklet, Sep. 2011.
- [91] D. Ä. Vamvuka, "Bio-oil, solid and gaseous biofuels from biomass pyrolysis processes - An overview," *International Journal of Energy Research*, no. 35, pp. 835–862, 2011.

- [92] E. D. Larson, "Biomass Gasification Systems for Electric Power, Cogeneration, Liquid Fuels, and Hydrogen," GCEP Biomass Energy Workshop, Stanford University, 2004.
- [93] "Alternative Fuels Data Center." [Online]. Available: http://www.afdc.energy.gov/fuels/fuel\_properties.php. [Accessed: 05-Apr-2013].
- [94] J. K. Kaldellis and K. A. Kavadias, "Cost-benefit analysis of remote hybrid winddiesel power stations: Case study Aegean Sea islands," *Energy Policy*, vol. 35, no. 3, pp. 1525–1538, Mar. 2007.
- [95] "Perkins Diesel Generator 200 kW" [Online]. Available: http://www.generatorsales.com/order/Perkins-200-kw-diesel-generator.asp?page=9 159.
   [Accessed: 08-May-2013].
- [96] Natural Resources Canada, "Average Wholesale (Rack) Prices for Diesel in 2012 Canada," [Online]. Available: http://www2.nrcan.gc.ca/eneene/sources/pripri/wholesale\_bycity\_e.cfm?PriceYear =2012&ProductID=13&LocationID=66&Average=3&dummy=#PriceGraph. [Accessed: 08-May-2013].
- [97] Natural Resources Canada, "Average Retail Prices for Auto Propane in Canada," 31-Oct-2012. [Online]. Available: http://www2.nrcan.gc.ca/eneene/sources/pripri/prices\_byyear\_e.cfm?ProductID=6. [Accessed: 01-May-2013].
- [98] T. A. Semelsberger, R. L. Borup, and H. L. Greene, "Dimethyl ether (DME) as an alternative fuel," *Journal of Power Sources*, vol. 156, no. 2, pp. 497–511, Jun. 2006.
- [99] "Density of Methanol from Dortmund Data Bank" [Online]. Available: http://www.ddbst.com/en/EED/PCP/DEN\_C110.php. [Accessed: 13-Aug-2013].
- [100] M. Boucher, A. Chaala, and C. Roy, "Bio-oils obtained by vacuum pyrolysis of softwood bark as a liquid fuel for gas turbines. Part I: Properties of bio-oil and its blends with methanol and a pyrolytic aqueous phase," *Biomass and Bioenergy*, vol. 19, no. 5, pp. 337–350, Nov. 2000.
- [101] "Methanex Methanol Price" [Online]. Available: http://www.methanex.com/products/methanolprice.html. [Accessed: 07-Aug-2013].

- [102] Canadian Gas Association, "Canadian Gas Association Gas Stats" [Online]. Available: http://www.cga.ca/resources/gas-stats/. [Accessed: 11-Sep-2013].
- [103] R. G. Lemus and J. M. Martínez Duart, "Updated hydrogen production costs and parities for conventional and renewable technologies," *International Journal of Hydrogen Energy*, vol. 35, no. 9, pp. 3929–3936, May 2010.

# **Appendix A: Additional calculations**

# Converting forest residue volume to mass

For a given volume of green wood, h:

$$h = V_{H_2O} + V_{ODW} \tag{A.1}$$

where  $V_{H_2O}$  and  $V_{ODW}$  are the volume of water and oven-dry wood within the total volume, respectively. If the green wood has a moisture content, *z*, defined as a percentage mass of the green wood, then Equation A.1 becomes:

$$h = \frac{(z - z_{ODW})M_{GT} \cdot 10^3}{\rho_{H_2O}} + \frac{\left(1 - (z - z_{ODW})\right)M_{GT} \cdot 10^3}{\rho_{ODW}}$$
(A.2)

where  $z_{ODW}$  is the moisture content of oven-dry wood,  $\rho_{H_2O}$  is the density of water and  $\rho_{ODW}$  is the density of oven-dry wood. A factor of 10<sup>3</sup> is introduced as  $M_{GT}$  is provided in tonnes rather than kilograms. Rearranging, Equation A.2 becomes:

$$M_{GT} = \frac{h}{10^3} \left( \frac{z - z_{ODW}}{\rho_{H_2O}} + \frac{(1 - z) + z_{ODW}}{\rho_{ODW}} \right)^{-1}$$
(3.1)

#### Weight and volume limits of trucks

Equation A.3 is used to determine if trucks that transport commodity j are limited by weight or by volume:

$$x = \left(\rho_j \times \nu_j\right) / w_j \tag{A.3}$$

where  $\rho_i$  is the density of commodity *j* (either woodchip, bio-oil, bio-slurry, or torrefied wood), and  $v_j$  and  $w_j$  are the volume and weight limit of the truck used to carry commodity *j*, respectively.

If x < 1, then the limiting factor is volume and if x > 1 the limiting factor is weight.

# **Bio-fuel quantity conversions**

Equations A.4 and A.5 detail the conversion between mass (kg) and volume (litre) or energy content (GJ) of bio-fuel m produced from commodity feedstock j at bio-fuel production facility k, respectively:

$$F_{litre,j,k,m} = \frac{F_{kg,j,k,m}}{\rho_m} \cdot 10^3 \tag{A.4}$$

$$F_{GJ,j,k,m} = \frac{F_{kg,j,k,m}}{l_m} \cdot 10^{-3}$$
(A.5)

where  $\rho_m$  is the density of bio-fuel m (not applicable for hydrogen) and  $l_m$  is the lower heating value of bio-fuel m.

# Diesel consumption at mobile torrefaction facilities

Diesel consumption at mobile torrefaction facilities is based on a 200 kW generator, which is capable of meeting maximum electrical demands of the mobile facility. However, depending on the capacity of the mobile torrefaction facility, electrical demands may decrease and diesel consumption will reduce. The rate at which diesel is consumed in based on data shown in Figure A-1.



Figure A-1 Diesel consumption rates for a 200 kW generator. [Data taken from [69]].

# Appendix B: Input data

All costs are in 2012 US\$.

# Forest residue resource

**Table B-1** Forest residue resource input data (technical)

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Annual volume of biomass	h	$m^3 \cdot yr^{-1}$	User	-	-
			defined		
Density of forest residues	$\varphi$	$m^3 \cdot km^{-2}$	65	-	[84]
Initial moisture content of forest residues	Ζ	%	50	40 - 60	[8], [70]
Moisture content of oven- dry wood (12% moisture)	z <sub>oDW</sub>	%	12	-	[8]
Density of oven-dry wood (12% moisture)	ρ <sub>odw</sub>	kg⋅m <sup>-3</sup>	500	400 - 865	[8]
Density of water	$\rho_{H_2O}$	kg⋅m <sup>-3</sup>	1000	-	-
Density of woodchips	$\rho_{WC}$	kg⋅m <sup>-3</sup>	300	220 - 350	[7], [8]
Energy required to dry woodchips	e <sub>dry</sub>	$MJ \cdot kg_{H^{2O}}^{-1}$	2.572	-	[68]
Lower heating value of oven-dry wood (12% moisture)	l <sub>oDW</sub>	MJ·kg <sup>-1</sup>	18	18 - 21	[23], [70]

**Table B-2** Forest residue resource input data (financial)

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Purchasing forest residues cost	C <sub>purchase</sub>	\$- GT <sup>-1</sup>	3.22	-	[49]
Piling forest residues cost	C <sub>piling</sub>	$GT^{-1}$	1.64	-	[85]
Chipping forest residues cost	C <sub>chipping</sub>	\$∙GT <sup>-1</sup>	13.73	-	[21]
# Transport

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Woodchip truck volume	v <sub>truck,WC</sub>	m <sup>3</sup>	70	-	[7]
Woochip truck weight limit	m <sub>truck,WC</sub>	tonne	21.5	-	[7]
Liquid B-train tanker	$v_{truck,liq}$	$m^3$	60	-	[79]
volume					
Liquid B-train tanker	m <sub>truck,lig</sub>	tonne	62.5	-	[78]
weight limit	-				
Torrefied wood truck	v <sub>truck,TW</sub>	m <sup>3</sup>	200	-	[78]
volume					
Torrefied wood truck	m <sub>truck.TW</sub>	tonne	62.5	-	[78]
weight limit					
Road tortuosity	τ	-	1.5	1.2 - 3	[63]
Additional transport	d	km	User	-	-
distance			defined		

## Table B-3 Transport input data (technical)

## Table B-4 Transport input data (financial)

		-			
Data	Symbol	Unit	Value	Cited Range	Reference(s)
Woodchip loading and	C <sub>tran,fix,WC</sub>	$GT^{-1}$	5.83	-	[86]
unloading	-				
Woodchip hauling	C <sub>tran,var,WC</sub>	\$·km <sup>-1</sup>	1.5	-	[87], [88] <sup>a</sup>
Bio-oil loading and	C <sub>tran,fix,BO</sub>	\$-tonne <sup>-1</sup>	5.55	-	[79]
unloading	-				
Bio-oil hauling	C <sub>tran,var,BO</sub>	$\cdot km^{-1}$	2.5	2.04 - 3.16	[88]
Bio-slurry loading and	C <sub>tran,fix,BS</sub>	\$-tonne <sup>-1</sup>	5.55	-	Ь
unloading					
Bio-slurry hauling	C <sub>tran,var,BS</sub>	\$·km <sup>-1</sup>	2.5	2.04 - 3.16	[88]
Torrefied wood loading	C <sub>tran,fix,TW</sub>	\$-tonne <sup>-1</sup>	5.83	-	с
and unloading	,,,				
Torrefied wood hauling	C <sub>tran.var.TW</sub>	$\cdot km^{-1}$	2.5	2.04 - 3.16	[88]

<sup>a</sup> trucking costs in British Columbia are on average 5% higher than in Alberta [88]

<sup>b</sup> assumed the same as bio-oil loading and unloading

<sup>c</sup> assumed the same as woodchip loading and unloading

## **Mobile facilities**

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Mobile facility size	S <sub>mob</sub>	ODTPD	50	-	[10]
Maximum capacity factor	$\sigma_{mob}$	%	87.5	-	[10]
Electricity requirement	β	GJ-ODT <sup>-1</sup>	0.3448	-	[10]
Diesel generator size	Sgen	kW	200	-	-
Diesel consumption rate	r <sub>diesel</sub>	litre hr <sup>-1</sup>	55		[69]
Bio-oil electricity generator efficiency	$\eta_{gen}$	%	30	-	[10]
Biomass dryer efficiency	$\eta_{dry}$	%	72	54 - 89	[68]
Staff required per facility	n <sub>staff</sub>	-	3	-	[10]
Density bio-oil	$ ho_{BO}$	kg∙m <sup>-3</sup>	1100	1100 - 1300	[11], [36], [38]
Bio-oil aqueous fraction	m <sub>WS</sub>	%wt	70	60 - 80	[17]
Density bio-slurry (30% char loading)	$ ho_{BS}$	kg⋅m <sup>-3</sup>	1300	-	[38]
Bio-slurry maximum char loading	$\zeta_{max}$	%wt	20	-	[38]
Density torrefied wood	$ ho_{TW}$	kg⋅m <sup>-3</sup>	250	180 - 300	[43]
Stainless steel tank reference size	S <sub>ref,tank</sub>	m <sup>-3</sup>	9400	-	[8]
Density propane	$\rho_{C_3H_8}$	kg⋅m <sup>-3</sup>	493	-	[89]
LHV bio-char	l <sub>BC</sub>	MJ·kg <sup>-1</sup>	30	28 - 31	[38], [90]
LHV bio-oil	l <sub>BO</sub>	MJ·kg <sup>-1</sup>	17	-	[29]
LHV syngas	$l_S$	MJ·kg <sup>-1</sup>	8	7.7 - 10	[91], [92]
LHV torrefied wood	$l_{TW}$	MJ·kg <sup>-1</sup>	20	18 - 23	[13]
LHV propane	$l_{C_3H_8}$	MJ·kg <sup>-1</sup>	46.35	-	[93]

 Table B-5 Mobile facility input data (technical)

#### Table B-6 Mobile facility input data (financial)

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Capital cost (per facility)	C <sub>mob</sub>	\$million	3.669	-	[10]
Operation and maintenance	0 <sub>mob</sub>	%capital	1.5	-	[10]
Cost per facility relocation	C <sub>relocate</sub>	\$	866	-	[10]
Staff wage	C <sub>wage,mob</sub>	\$• yr <sup>-1</sup>	120,765	-	[10]
Diesel generator cost	C <sub>gen,mob</sub>	\$	50,000	-	[94], [95]
Stainless steel tank	C <sub>ref,tank</sub>	\$million	1.0	-	[8]
reference cost					
Stainless steel tank	0 <sub>tank</sub>	%capital	2	-	-
maintenance					
Stainless steel tank cost	κ <sub>tank</sub>	-	0.65	-	[79]
scaling factor					
Diesel purchase price	C <sub>diesel</sub>	\$·litre <sup>-1</sup>	0.85	-	[96]
Propane purchase price	$C_{C_3H_8}$	\$•litre <sup>-1</sup>	0.7	-	[97]

## **Bio-fuel facilities**

Table B-7 Bio-fuel	facility in	put data (	(technical)
--------------------	-------------	------------	-------------

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Gasification and Fischer-	S <sub>ref.FT</sub>	ODTPD	2000	-	[49]
Tropsch facility reference					
size					
Gasification and Fischer-	$f_{FT}$	% wt feed	12.96	-	[49]
Tropsch facility bio-fuel					
production					
Gasification and Fischer-	$f_{r,FT}$	-	0.4	-	[49]
Tropsch facility bio-fuel					
production ratio (petrol					
/diesel)					
Gasification and water gas	S <sub>ref,WGS</sub>	ODTPD	2000	-	[48], [49]
shift facility reference size					
Gasification and water gas	<i>f<sub>WGS</sub></i>	% wt feed	7.5	6.8 - 8.0	[48]
shift facility bio-fuel					
production					
Upgrading bio-oil facility	S <sub>ref,UG</sub>	TPD	1500	-	[16]
reference size					
Upgrading bio-oil facility	f <sub>UG</sub>	% wt feed	33	-	[16]
bio-fuel production					
Upgrading bio-oil facility	$f_{r,UG}$	-	0.95	-	[16]
bio-fuel production ratio					
(petrol/diesel)			_	<b>a</b> 10 <b>a a</b>	54 63 57 43
Upgrading bio-oil facility	$\gamma_{H_2}$	%wt	5	2.18 - 5.39	[16], [71],
hydrogen requirement		b10-01l	<b>7</b> 4 400		[72]
SMR facility for hydrogen	S <sub>ref,SMR</sub>	$kg_{CH4} \cdot day^{-1}$	74,400	-	[16]
production	6	<b>0</b> (	0.44		51.63
SMR hydrogen production	<i>İ<sub>SMR</sub></i>	% wt <sub>CH4</sub>	0.44	-	[16]
Steam reformation of bio-	S <sub>ref,SR</sub>	TPD	1198	-	[19]
oil facility reference size					
Steam reformation of bio-	f <sub>sr</sub>	%wt	14.7	14.7 - 17.2	[19]
oil facility bio-fuel		bio-oil			
production		(WS)	10		54.03
Quantity of methanol	$m_{meth}$	%wt	10	-	[19]
added to bio-oil		b10-01	0.5.6		50.01
Density diesel	$ ho_D$	kg·m <sup>°</sup>	856	-	[98]
Density petrol	$ ho_P$	kg·m⁻³	737	-	[98]
Density methanol	$ ho_{meth}$	kg⋅m <sup>-3</sup>	791.3	-	[99]
Density bio-oil/methanol	$\rho_{BO-meth}$	kg⋅m <sup>-3</sup>	1040	-	[100]
blend (10% wt methanol)					
LHV diesel	$l_D$	MJ·kg <sup>-1</sup>	41.66	-	[98]
LHV petrol	$l_P$	MJ·kg <sup>-1</sup>	43.47	-	[98]
LHV hydrogen	$l_H$	MJ·kg <sup>-1</sup>	120	-	[98]
LHV methane	l <sub>CH4</sub>	MJ·kg <sup>-1</sup>	47.79	-	[98]
Build time	$t_b$	yr	0	-	-

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Operation lifetime	$t_{op}$	yr	20	-	-
Capacity factor (all bio-fuel facilities)	$\sigma_{bf}$	%	90	-	-

 Table B-8 Bio-fuel facility input data (financial)

Data	Symbol	Unit	Value	Cited Range	Reference(s)
Gasification and Fisher-	$C_{FT}$	\$million	269.02	-	[49]
Tropsch facility reference	11				
cost					
Gasification and Fischer-	0 <sub>hf.FT</sub>	%	6	-	[49]
Tropsch facility operation	<i>b</i> ), i i				
and maintenance					
Gasification and water gas	C <sub>WGS</sub>	\$million	255.15	255 - 370	[48], [49]
shift facility reference cost	in de				
Gasification and water gas	O <sub>bf.WGS</sub>	%	6	-	[49]
shift facility operation and	<i>b)</i> ,,,, db				
maintenance					
Feed preparation	C <sub>screen</sub>	\$million	1.90	-	[68]
equipment: Screening	5010011				
(2000 TPD bio-fuel					
facility)					
Feed preparation	C <sub>drv</sub>	\$million	9.43	-	[68]
equipment: Rotary dryer					
(2000 TPD bio-fuel					
facility)					
Feed preparation	C <sub>arind</sub>	\$million	12.10	-	[68]
equipment: Grinding (2000	<i>g</i>				
TPD bio-fuel facility)					
Upgrading bio-oil facility	C <sub>UG</sub>	\$million	138.34	-	[16]
reference cost		- ,			-4 -7
Upgrading bio-oil facility	O <sub>bf,UG</sub>	%	2	-	[16]
operation and maintenance					
SMR facility for hydrogen	C <sub>SMR</sub>	\$million	95.94	-	[16]
production reference cost					
SMR facility for hydrogen	O <sub>bf,SMR</sub>	%	2	-	[16]
production operation and					
maintenance			1.5.5.00		54.03
Steam reformation of bio-	C <sub>SR</sub>	\$million	155.00	-	[19]
oil facility reference cost					
Steam reformation of bio-	O <sub>bf,SR</sub>	%	4	-	-
oil facility operation and					
maintenance		+ 3			
Open air pile storage	C <sub>pile</sub>	\$•m <sup>-3</sup>	1.31	-	[73]
Cost of methanol	C <sub>meth</sub>	\$•kg <sup>-1</sup>	0.53	-	[101]
Cost of natural gas	C <sub>CH4</sub>	$GJ^{-1}$	5.24	-	[102]
Cost of hydrogen from	C <sub>external-H<sub>2</sub></sub>	\$•kg <sup>-1</sup>	5	1.5 - 10	[71], [103]
external source	-				

Me C-1 Sensitivity study results for levelised delivered cost	signifies result change of less than 10%
Table C-	(-) signifi

Input parameter	Wood	chip	Bio-	oil	Bio-slu	ırry	Torrefie	d Wood
	-50%	+50%	-50%	+50%	-50% +	-50%	-50%	+50%
		Percent	tage char	ige in Le	velised D	elivered	l Cost	
Bio-oil density	I	ı	11.0	1	I	I	1	1
Mobile facility reference cost	I	I	-10.5	10.5	-10.2	10.2	-11.6	11.6
Volume of mobile product transport truck	I	ı	I	ı	I	I	22.9	I
Density of torrefied wood	ı	ı	ı	·	ı	ı	22.9	I
Weight limit of mobile product transport truck	ı	ı	12.2	ı	14.7	ı		
Char loading of bio-slurry	ı	ı	ı	ı	18.4	-16.1	I	1
Mobile facility electricity generator efficiency	I	ı	14.2	ı	13.7	I	I	I
Cost of propane	1	'	-10.3	10.3	ı	1	1	ı
Haul cost factor of B train trucks	ı	'	ı	ı	ı	'	-11.5	11.5
Haul cost factor of woodchip truck	-36.3	36.3	ı	·	ı	ı	I	I
Woodchip density	72.5	'	ı	'	ı	'	ı	ı
Volume of woodchip truck	72.5	ı	ı	ı	ı	ı	ı	ı
Tortuosity factor	-36.3	36.3	ı	'	ı	ľ	-12.8	12.8
Dryer efficiency	ı	'	45.5	-15.2	43.9	-14.6	45.3	ı
Initial moisture content of forest residues	-28.7	67.6	-29.1	101.7	-28.1	98.3	-9.5	107.3

# Appendix C: Sensitivity study results

Table C-2 Sensitivity study results for levelised cost of bio-fuel (Gasification and Fischer-Tropsch bio-fuel production route)

(-) signifies result change of less than 10%

Input parameter	Woodd	chip	Bio-c	lid	Bio-slı	irry	<b>Forrefied</b>	Wood
	-50% +	+50%	-50% -	-50%	-50% -	-50%	-50% +	-50%
		Percent	age chang	ge in Lev	velised Co	ost of Bi	o-fuel	
Volume of mobile product transport truck	ı	ı	I	ı	ı	ı	15.5	'
Density of torrefied wood	'	ı	ı	ı	ı	ı	17.3	ı
Bio-oil upgrading hydrogen requirement	1	ı	ı	ı	'	ı	•	ı
Weight limit of mobile product transport truck	'	ı	I	ı	10.9	ı	ı	I
Reference cost of bio-fuel facility	-15.8	15.8	-12.6	12.6	-13.2	13.2	-14.6	14.6
Haul cost factor of woodchip truck	-22.7	22.7	ı	ı	ı	ı	ı	I
Volume of woodchip truck	45.3	I	I	I	I	I	ı	I
Mobile facility electricity generator efficiency	ı	ı	12.0	ı	11.7	ı	ı	I
Woodchip density	48.4	ı	I	I	I	ı	ı	I
Aqueous fraction of bio-oil	'	ı	I	ı	I	ı	ı	I
Bio-hydrogen produced from bio-oil steam reformation	ı	ı	I	I	I	I	ı	I
Tortuosity factor	-22.7	22.7	ı	ı	ı	ı	ı	I
Discount factor	'	10.3	-10.6	12.0	-10.4	11.7	-12.1	13.6
Bio-fuel production factor	100.0	-33.3	100.0	-33.3	100.0	-33.3	100.0	-33.3
Dryer efficiency	'	1	33.3	-11.1	32.3	-10.8	30.7	ı
Initial moisture content of forest residues	-21.4	49.5	-23.2	78.3	-22.7	76.3	'	77.2

Table C-3 Sensitivity study results for levelised cost of bio-fuel (Gasification and water gas shift bio-fuel production route)

(-) signifies result change of less than 10%

Input parameter	Woodc	thip	Bio-6	lic	Bio-slı	, irry	<b>Forrefied</b>	Wood
	-50% +	-50%	-50% -	⊦50%	-50% -	+50%	-50% +	-50%
1		Percenta	ige chang	ge in Lev	relised Co	ost of Bi	o-fuel	
Volume of mobile product transport truck	I	I	ı	I	I	I	15.8	1
Density of torrefied wood	ı	ı	ı	'	ı	ı	17.6	1
Bio-oil upgrading hydrogen requirement	ı	ı	ı	ı	ı	ı	'	ľ
Weight limit of mobile product transport truck	I	ı	ı	ı	11.0	I	ı	I
Reference cost of bio-fuel facility	-15.3	15.3	-12.1	12.1	-12.7	12.7	-14.0	14.0
Haul cost factor of woodchip truck	-23.0	23.0	ı	ı	I	ı	'	ı
Volume of woodchip truck	46.1	ı	ı	ı	I	I	ı	I
Mobile facility electricity generator efficiency	ı	'	12.0	'	11.8	ı	'	ı
Woodchip density	49.2	ı	ı	ı	I	I	ı	I
Aqueous fraction of bio-oil	ı	ı	ı	'	ı	ı	'	ı
Bio-hydrogen produced from bio-oil steam reformation	ı	ı	ı	ı	ı	ı	'	ı
Tortuosity factor	-23.0	23.0	ı	ı	ı	ı	·	ı
Discount factor	ı	10.0	-10.5	11.9	-10.2	11.5	-11.9	13.5
Bio-fuel production factor	100.0	-33.3	100.0	-33.3	100.0	-33.3	100.0	-33.3
Dryer efficiency	ı	ı	33.7	-11.2	32.8	-10.9	31.1	ı
Initial moisture content of forest residues	-21.7	50.0	-23.4	79.1	-22.9	77.1	•	78.2

Table C-4 Sensitivity study results for levelised cost of bio-fuel (Upgrading bio-oil bio-fuel production route) (-) signifies result change of less than 10%

Input parameter	SMI	~	BOS	SR	Purch	ase
	-50% -	F50%	-50%	+50%	-50% -	+50%
	Percenta	ige chan	ge in Lev	/elised C	ostofBio	o-fuel
Volume of mobile product transport truck	ı	ı	ı	ı	ı	I
Density of torrefied wood	ı	ı	ı	I	'	ı
Bio-oil upgrading hydrogen requirement	'	ı	-17.5	17.2	'	'
Weight limit of mobile product transport truck	ı	ı	I	I	•	I
Reference cost of bio-fuel facility	1	ı	ı	I	•	ı
Haul cost factor of woodchip truck	ı	ı	ı	I	ı	I
Volume of woodchip truck	ı	ı	ı	I	ı	I
Mobile facility electricity generator efficiency	11.1	ı	12.4	ı	10.4	I
Woodchip density	ı	ľ	I	I	ı	I
Aqueous fraction of bio-oil	'	ı	34.2	-11.6	'	'
Bio-hydrogen produced from bio-oil steam reformation	'	ı	34.2	-11.6	ı	'
Tortuosity factor	ı	'	ı	ı	'	ı
Discount factor	-11.2	12.7	-10.8	12.1	'	ı
Bio-fuel production factor	100.0	-33.3	100.0	-33.3	100.0	-33.3
Dryer efficiency	31.1	-10.4	35.7	-11.9	31.1	-10.4
Initial moisture content of forest residues	-21.6	72.9	-24.4	83.0	-20.7	71.2

Table C-5 Sensitivity study results for levelised cost of bio-fuel (Steam reformation of bio-oil bio-fuel production route) (-) signifies result change of less than 10%

Input parameter	Bio-oil	
	-50% +50%	
Pe	rcentage change in Levelised Cost of Bio-fue	el
Volume of mobile product transport truck		'
Density of torrefied wood	ı	ī
Bio-oil upgrading hydrogen requirement		
Weight limit of mobile product transport truck	ı	ī
Reference cost of bio-fuel facility		ī
Haul cost factor of woodchip truck		ı
Volume of woodchip truck		ī
Mobile facility electricity generator efficiency	12.4	
Woodchip density		ī
Aqueous fraction of bio-oil	100.0 -33.3	ŝ
Bio-hydrogen produced from bio-oil steam reformation	100.0 -33.3	ŝ
Tortuosity factor	I	
Discount factor	-10.3 11.6	9
Bio-fuel production factor	100.0 -33.3	ŝ
Dryer efficiency	36.1 -12.0	0
Initial moisture content of forest residues	-24.5 83.6	9